# Migrants, Ancestors, and Foreign Investments

# KONRAD B. BURCHARDI

Institute for International Economic Studies, Stockholm University, BREAD and CEPR

# THOMAS CHANEY

Sciences Po

TAREK A. HASSAN

Boston University, NBER and CEPR

First version received May 2016; Editorial decision June 2018; Accepted August 2018 (Eds.)

We use 130 years of data on historical migrations to the U.S. to show a causal effect of the ancestry composition of U.S. counties on foreign direct investment (FDI) sent and received by local firms. To isolate the causal effect of ancestry on FDI, we build a simple reduced-form model of migrations: Migrations from a foreign country to a U.S. county at a given time depend on (1) a push factor, causing emigration from that foreign country to the entire U.S., and (2) a pull factor, causing immigration from all origins into that U.S. county. The interaction between time-series variation in origin-specific push factors and destination-specific pull factors generates quasi-random variation in the allocation of migrants across U.S. counties. We find that doubling the number of residents with ancestry from a given foreign country relative to the mean increases the probability that at least one local firm engages in FDI with that country by 4 percentage points. We present evidence that this effect is primarily driven by a reduction in information frictions, and not by better contract enforcement, taste similarities, or a convergence in factor endowments.

Key words: Migrations, Foreign directinvestment, International trade, Networks, Social ties.

JEL Codes: O11, J61, L14

International migrations have reached unprecedented levels over the past decades,<sup>1</sup> shaping an increasingly ethnically diverse and socially connected world. The economic consequences of these migrations are at the heart of fierce political debates on immigration policy, yet our understanding of their economic effects remain incomplete. At the same time, foreign direct investment (FDI) has become a defining feature of international production.<sup>2</sup> Local policymakers

1. In 2013, there were 232 million international migrants, an all time high (UN Population Facts No. 2013/2).

2. In 2009, 55% of all U.S. exports emanated from U.S. multinationals that operated subsidiaries abroad. These firms employ 23 million Americans, while U.S. subsidiaries of foreign firms employ another 5 million. *Source*: Office of the United States Trade Representative, Fact Sheet on International Investment.

The editor in charge of this paper was Gita Gopinath.

#### **REVIEW OF ECONOMIC STUDIES**

see attracting and retaining FDI as a major goal, and technology transfers through FDI are both a conduit for technological progress abroad and a source of revenue for U.S. firms.<sup>3</sup> Migrations and FDI create two parallel global networks, one of ethnic connections, one of parent–subsidiary linkages. How do these two networks affect each other? In this article, we estimate the long-term effect of immigration on the patterns of FDI sent and received by U.S. firms, and shed light on the mechanism behind this effect. We show the ethnic connections emanating from migrations reaching back more than a century have a large positive causal effect on the propensity of U.S. firms to engage in FDI with the historical migrants' countries of origin; and this effect appears to be driven primarily by a reduction of information frictions.

Evaluating the causal impact of migrations on FDI requires a rigorous identification strategy, as unobserved factors may simultaneously affect migrations, ancestry, and FDI, creating a spurious correlation between them. For example, historical migrations might have occurred between origins and destinations with certain unobserved climatic or other characteristics, and these characteristics might in turn drive FDI today. Similarly, past FDI might itself cause migration, for example because foreign employers second employees to work at their U.S. subsidiaries.

To overcome these challenges, we construct a set of instrumental variables (IV) for the presentday ancestry composition of U.S. counties, best explained by the examples of migrations from Germany and Italy. German migrations peaked at the end of the nineteenth century when the Midwest was booming and attracting large numbers of migrants. We observe a large population with German ancestry in the Midwest today. Italian migrations peaked a few decades later, at the beginning of the twentieth century when the West was attracting large numbers of migrants. We observe a large population with Italian ancestry in the West today. We use this interaction of time-series variation in the relative attractiveness of different destinations within the U.S. (*e.g.* end of nineteenth century Midwest versus early twentieth century West) with the staggered arrival of migrants from different origins (*e.g.* end of nineteenth century Germany versus early twentieth century Italy) to instrument for the present-day distribution of ancestries. This formal IV strategy is essential. For instance, while the effect of ancestry on FDI is positive in both ordinary least squares (OLS) and IV specifications, its effect on international trade drops and becomes insignificant when we instrument for ancestry, suggesting that unobservable factors indeed confound simple OLS estimates of these effects.

Our article makes three main contributions: (1) historical migrations and the ethnic diversity they created have a quantitatively large causal effect on FDI; (2) this ethnic determinant of FDI appears to operate mainly because it facilitates the flow of information; and (3) we propose a general method for instrumenting the composition of ancestry across U.S. counties.

Before describing the related literature, we summarize our main empirical results.

We find that, for an average U.S. county, doubling the number of individuals with ancestry from a given origin country increases by 4 percentage points the probability that at least one firm from this U.S. county engages in FDI with that country, and increases by 7% the number of local jobs at subsidiaries of firms headquartered in that country. These effects persist over generations: even nineteenth century migrations significantly affect the patterns of FDI today.

To arrive at those findings on the *causal* impact of foreign ancestry on the patterns of FDI, we follow an IV strategy. We motivate our approach using a simple reduced-form dynamic model of migrations. Migrations from a given origin country o to a given U.S. destination county d in period t depend on the total number of migrants arriving in the U.S. from o (a push factor), the relative economic attractiveness of d to migrants arriving in t (a pull factor), and the size of the pre-existing local population of ancestry o in d at t, allowing for the fact that migrants

<sup>3.</sup> See McGrattan and Prescott (2010) and Holmes et al. (2015).

tend to prefer settling near others of their own ethnicity (a recursive factor). Solving the model recursively shows the number of residents in d today who are descendants of migrants from o is a function of simple and higher-order interactions of the sequence of pull and push factors.

To construct valid instruments from this sequence of interactions, we isolate variation in the pull and push factors that is plausibly independent of any unobservables that may make a given destination within the U.S. differentially more attractive for FDI from a given origin country. To that end, we measure the pull factor from country o to county d as the fraction of migrants coming from anywhere in the world who settle in d at time t, excluding migrants from the same continent as o. The pull from o towards d thus depends only on the destination choices of migrants arriving at the same time from other continents. Similarly, we measure the push factor as the total number of migrants arriving in the U.S. from o at time t, excluding migrants from o who settled in the same region as d. We then instrument for the present-day number of residents in county d with ancestry from country o using the full set of simple and higher-order interactions of these pull and push factors. Using the entire series of interactions going back to 1880 maximizes the statistical power of our IV strategy.

A major advantage of this "leave-out push-pull" approach is that it yields a specific instrument for migrations from each origin to each destination at each point in time, uniquely allowing us to simultaneously control for both origin and destination fixed effects, and to conduct a number of falsification exercises and robustness checks. For example, we obtain quantitatively very similar effects of ancestry on FDI when we combine our IV strategy with a natural experiment surrounding the rise and fall of communism. These specifications, similar to a difference-indifference, measure how variations in ancestry driven only by the instrumented inflow of defectors from communist countries explain changes in FDI, from zero in 1989 to its current level in 2014.

This flexibility of our instruments also delivers the statistical power to isolate specific channels linking ancestry to FDI: common ancestry may affect FDI because it (1) induces similarities in tastes for consumption, (2) causes a convergence in factor endowments, facilitating horizontal FDI, (3) provides social collateral for contract enforcement, substituting for poor institutions, or (4) reduces information frictions. We find no evidence in support of the first three channels. Common ancestry does not affect FDI in the final goods sector more than in the intermediate goods sector, does not appear to cause a convergence in the sectoral distribution of employment, and has a significantly weaker impact on FDI for countries with weak institutions.

To provide a direct test for the remaining hypothesis that common ancestry affects FDI by reducing information frictions, we construct a novel measure of information demand about foreign countries using data from Google internet searches. Our index reflects variation across U.S. metro areas in the relative frequency of search terms containing the names of each countries' most prominent politicians, actors, athletes, and musicians. We find a large causal effect of common ancestry on this index: residents of U.S. metro areas with more ancestry from a given country systematically acquire more information about the politics and culture of that country. This fact fully accounts for the effect of ancestry on FDI, in the sense that controlling for our index of information demand drives out the significance of common ancestry in predicting FDI.

We also find that the effects of ancestry on FDI and information flow continue to operate long after migration from the origin country ceases, suggesting that immigrants pass traits to their descendants that facilitate economic exchange with their origin countries.

As one example, foreign ancestry increases the use of the origin country's language by U.S.-born individuals.

To illustrate the quantitative implications of our results, we conduct two thought experiments. In the first, we calculate the effect of Chinese exclusion — the effective ban on Chinese immigration between 1882 and 1965. Absent this ban, we predict the fraction of counties in the Northeast with FDI links to China would have increased substantially (e.g. doubled in New York

state). In the second, we calculate the effect of a hypothetical "L.A. gold rush" — an early population growth in Los Angeles before 1880 similar to the experience of San Francisco. We predict there would be 60,000 more individuals with German and Irish ancestry in Los Angeles today, and FDI between Los Angeles and Germany and Ireland would have increased by around 60%. The effect of ancestry on FDI is thus large and economically important.

Finally, we note one important limitation to our analysis: Our results rely purely on variation in the composition of FDI within the U.S., not between countries. Although we believe that, in light of our results, the ethnic diversity of the U.S. likely also raises FDI for the country as a whole, we cannot exclude the possibility that increases in FDI in one state are partially or fully offset by decreases in others.

**Existing literature.** A large literature shows that measures of affinity between regions, such as common ancestry, social ties, trust, and telephone volume, correlate strongly with aggregate economic outcomes, such as foreign direct investment (Guiso *et al.*, 2009; Leblang, 2010), international asset flows (Portes and Rey, 2005), and trade flows (Gould, 1994; Rauch and Trindade, 2002).<sup>4</sup> How much of this association should be interpreted as causal, however, remains an open question because these measures of affinity are likely to be non-random.

Three recent papers make attempts at identifying a causal effect of migrations on FDI and trade. Javorcik *et al.* (2011) use the cost of acquiring a passport and the existing stock of migrants from different countries in the U.S. to instrument for the impact of migrations between foreign countries and the U.S. on FDI. However, these instruments are most likely correlated with both migrations to the U.S. *and* FDI flows, and thus likely violate the exclusion restriction. Cohen *et al.* (2015) use the location of Japanese internment camps during World War II, and Parsons and Vezina (2018) the placement of Vietnamese refugees after the Vietnam War to identify a causal effect of those migrations on contemporary trade flows between locations within the U.S., and Japan and Vietnam, respectively. While the exclusion restriction for the instruments in those two papers is plausible, instrumenting for migrations from only one country makes it impossible to control for destination fixed effects, that is, unobserved factors making a U.S. state both a large recipient of migrants and a large importer and exporter.

Burchardi and Hassan (2013) use variation in wartime destruction across West German regions to show evidence of a causal effect of social ties on changes in GDP growth and FDI in East Germany after the fall of the Berlin Wall.<sup>5</sup> Redding and Sturm (2008), Juhász (2018), and Steinwender (2018) study the effect of historical shocks on economic interactions across borders.

We contribute to this literature in several ways. First, we identify a causal effect of ancestry on FDI in a setting with a high degree of external validity directly relevant for assessing, for example, the long-term effects of immigration policy. Second, because our identification strategy can be applied to all origin countries and destination U.S. counties, we are able to guard against a wide range of possible confounding factors and to relate to the previous literature by employing a gravity equation with both destination and origin fixed effects. Third, we show that ancestry affects FDI most likely due to its effects on information flow.

Our article also contributes to the debate on the costs and benefits of immigration. Much of the existing literature has focused on the effects of migration on local labour markets, mostly in the short run.<sup>6</sup> A more recent literature focuses on the effect of cultural, ethnic, and birthplace

<sup>4.</sup> Also see Head and Ries (1998), Combes *et al.* (2005), Garmendia *et al.* (2012), and Aleksynska and Peri (2014) for the relationship between common ancestry and trade and Bhattacharya and Groznik (2008) for its relationship with FDI.

<sup>5.</sup> See Fuchs-Schündeln and Hassan (2015) and Chaney (2016) for surveys of this literature.

<sup>6.</sup> See, for example, Card (1990), Card and Di Nardo (2000), Friedberg (2001), Borjas (2003), and Cortes (2008). Borjas (1994) provides an early survey.

diversity on economic development and growth.<sup>7</sup> Most closely related are Nunn *et al.* (2017) who study the effect of immigration from all origins during the Age of Mass Migration on presentday outcomes. Fulford *et al.* (2015) study the effect of historical ancestry composition of U.S. counties on local economic growth. We add to this literature by showing a long-term effect of migration on the absolute advantage in conducting FDI of different regions that may explain part of the association between diversity and long-term growth found in other studies.

Our approach to identification is related to Card (2001) who instruments immigration flows from origin o to destination d with the interaction of the total immigration from o to the U.S. (the push factor) and the spatial distribution of previous migrants from o in the U.S. (the recursive factor). This strategy however is not appropriate in our context, where unobserved and persistent origin-destination specific characteristics (such as the local climate) may drive both the spatial distribution of previous migrants and FDI. Our approach instead combines a push-pull model similar to that of Card (2001) with a two-dimensional version of the leave-out approach of Bartik (1991) and Katz and Murphy (1992), and uses historical migrations going back to the nineteenth century to instrument for the current stock of ancestry. This hybrid approach can easily be replicated for other countries, time periods, or any variable where cumulated flows matter, without the need for a rare or even unique historical accident.

The remainder of this article is structured as follows. Section 1 introduces our data. Section 2 gives a brief overview of the history of migration to the U.S. Section 3 identifies the causal effect of ancestry composition on FDI, conducts robustness checks, and illustrates the quantitative implications of our findings using two thought experiments. Section 4 examines the mechanism underlying the effect of ancestry on FDI.

#### 1. DATA

We collect data on migrations and ancestry, on FDI and trade, and on origin and destination characteristics. Below is a description of our data, along with their source. Further details on the construction of all data are given in Supplementary Appendix A.

#### 1.1. Migrations and Ancestry

Our migration and ancestry data are constructed from the individual files of the Integrated Public Use Microdata Series (IPUMS) samples of the 1880, 1900, 1910, 1920, 1930, 1970, 1980, 1990, and 2000 waves of the U.S. census, and the 2006–2010 five-year sample of the American Community Survey. We weigh observations using the personal weights provided by these data sources. Supplementary Appendix Table 1 summarizes specific samples and weights used. We cannot use data from the 1940, 1950, and 1960 censuses, because these did not collect information on the year of immigration. The original 1890 census files were lost in a fire.

Throughout the article, we use t-1 and t to denote two consecutive census waves, o for the foreign country of origin, and d for the U.S. destination county. We construct the number of migrants from origin o to destination d at time t,  $I_{o,d}^t$ , by counting the number of respondents who live in d, were born in o, and emigrated to the U.S. between t-1 and t. The exception to this rule is the 1880 census (the first in our sample), which also did not record the year of immigration. The variable  $I_{o,d}^{1880}$  instead measures the number of residents who were either born in o or whose parents were born in o, thus covering the two generations of immigrants arriving

<sup>7.</sup> See Ottaviano and Peri (2006), Putterman and Weil (2010), Peri (2012), Ashraf and Galor (2013), Ager and Brückner (2013), Alesina *et al.* (2016a), and Alesina *et al.* (2016b).

prior to 1880.<sup>8</sup> Since 1980, respondents have also been asked about their primary ancestry in both the U.S. Census and the American Community Survey, with the option to provide multiple answers. Ancestry<sup>t</sup><sub>o,d</sub> corresponds to the number of individuals residing in *d* at time *t* who report *o* as first ancestry. Note that this measure captures self-reported (recalled) ancestry.<sup>9</sup>

The respondents' residence is recorded at the level of historic counties, and at the level of historic county groups or PUMAs from 1970 onwards. Whenever necessary we use contemporaneous population weights to transition data from the historic county group or PUMA level to the historic county, and then use area weights to transition data from the historic county level to the 1990 U.S. county level.<sup>10</sup> The respondents' stated ancestry (birthplace) often, but not always, directly corresponds to foreign countries in their 1990 borders (*e.g.* "Spanish" and "Denmark"). When no direct mapping exists (*e.g.* "Basque" or "Lapland"), we construct transition matrices that map data from the answer level to the 1990 foreign country level, using approximate population weights where possible and approximate area weights otherwise. In the few cases when answers are imprecisely specific or such a mapping cannot be constructed (*e.g.* "European" or "born at sea"), we omit the data.<sup>11</sup> The resulting dyadic dataset covers 3,141 U.S. counties, 195 foreign countries, and 10 census waves.

#### 1.2. Foreign direct investment

Our data on FDI is from the U.S. file of the 2014 edition of the Bureau van Dijk ORBIS data set.<sup>12</sup> For each U.S. firm, the database lists the location of its (operational) headquarters, the addresses of its foreign parent entities, and the addresses of its partially or fully owned international subsidiaries and branches. In our main specification, we treat all equity stakes of any size as constituting a parent-subsidiary link.<sup>13</sup> Altogether, we have information on 36,108 U.S. firms that have at least one foreign parent or subsidiary. Collectively, these firms have 102,618 foreign parents and 176,332 foreign subsidiaries.<sup>14</sup> Our main outcome variable,  $1[FDI_{o,d} > 0]$ , is 1 if at least one firm in *d* has at least one parent or subsidiary in *o*, 0 otherwise. It captures both outward FDI (U.S. firms with foreign subsidiaries) and inward FDI (foreign firms with U.S. subsidiaries). We also count the number of FDI linkages between *o* and *d* (the number of foreign parents and subsidiaries in *o* of all firms in *d*), the number of unique parent in *o* (# *of Employees at Subsidiaries in Destination*),<sup>15</sup> and the two-digit NAICS code of the sector of the U.S. firm. See Supplementary Appendix A.2 for details. The resulting dataset covers the same 3,141 U.S. counties, 195 foreign countries, and 612,495 origin-destination pairs as above.

8. If the own birthplace is in the U.S., imprecisely specific (*e.g.* a continent), or missing, we instead use the parents' birthplace, assigning equal weights to each parent's birthplace.

9. See Duncan and Trejo (2016) for recent evidence on recalled versus factual ancestry in CPS data.

10. We also aggregate our data to the PUMA level and show that our results are robust.

11. Supplementary Appendix Tables 2 and 3 report summary statistics on these data transitions, including the share of affected respondents. Supplementary Appendix A.1 provides a detailed description of the data transformation.

12. In robustness checks we show that our results do not change when we instead use data from the 2007 file.

13. Supplementary Appendix Table 11 shows that our results are almost completely unchanged when we restrict ourselves to links with an ownership stake larger than 5%, 25%, or 50%.

14. Although Bureau van Dijk cross checks the data on international subsidiaries and branches using both U.S. and foreign data sources, we cannot exclude the possibility that coverage may be better for some countries than for others. However, all of our specifications control for country fixed effects such that any such variation in coverage at the country level would not affect our results.

15. When information on the number of employees is missing (which is the case for 95% and 58% of subsidiaries in the destination and origin, respectively), we assume the subsidiary employs one person.

#### 1.3. Other Data

To streamline the exposition, we discuss our measure of information demand in Section 4.2. In addition, we use data on aggregate trade flows between U.S. states and foreign countries for the year 2012 from the U.S. Census Bureau.<sup>16</sup> We construct geographic distances, absolute latitude differences, and measures of agricultural similarity between U.S. counties and foreign countries, and collect information on a number of characteristics for countries, counties, and sectors. See Supplementary Appendix A.3 for details.

## 1.4. Summary statistics

Panel A of Table 1 gives summary statistics on our sample of  $3,141 \times 195$  origin-destination pairs.<sup>17</sup> Column 1 shows means and standard deviations for all observations. Columns 3–4 show the same statistics for the subsamples of origin-destination pairs containing only observations with non-zero ancestry, and ancestry in the bottom and top quintile, respectively. The table shows that a lot of the variation both in ancestry and FDI is at the extensive margin. Only 1.8% of origin-destination pairs have an FDI link. Conditional on the U.S. county having any population with origins in the foreign country, 3.1% have an FDI link. The larger this population, the larger the probability of finding an FDI link, with 12.8% of the origin-destination pairs in the top quintile having an FDI link. Supplementary Appendix Figure 3 visualizes examples of this relationship: for each of the ten largest U.S. counties in terms of total FDI linkages, it scatters the number of FDI links to o against the number of individuals of ancestry o in d; regression lines are fitted for the top 10% ancestry groups and bottom 90% separately. In almost all cases larger ancestry groups are more likely to have an FDI link, and they have more FDI links. Panel A of Table 1 also shows that about half of the origin-destination pairs have ancestry of zero: most destinations in the U.S. do not have populations with ancestry from all 195 origin countries. The mean number of individuals with ancestry from a given origin is 316, but is highly skewed, with a mean in the top quintile of 2,852 individuals. Compared to this stock of ancestry, the flow of immigrants between 1990 and 2000 is relatively small, with twenty-three on average across the sample. The summary statistics also show that the number of first-generation immigrants (foreign born) measured in the 2010 American Communities Survey appears somewhat understated (sixty-nine on average). This fact is known in the literature and appears to affect only the measurement of immigration flows but not the stock of ancestry (Jensen et al., 2015). For this reason, we exclude the 2000–2010 wave of migrations from our standard specification (its inclusion however has no effect on any of our main results).

Panels B and C show summary statistics following the same format for destination counties and origin countries for variables used in our estimation of heterogenous effects. Supplementary Appendix Table 4 gives summary statistics on the intensive margin of FDI.

# 2. HISTORICAL BACKGROUND

The 1880 U.S. census counted 50 million residents, 10 million of which were first- or second-generation immigrants from 195 countries. The censuses taken since 1880 counted an additional 67 million immigrants. Our sample period thus covers the vast majority of migrations.<sup>18</sup>

16. When we aggregate our dataset across U.S. states, the correlation with aggregate trade between the entire U.S. and foreign countries from the NBER bilateral trade dataset is 99.9% for imports and 99.7% for exports respectively (in 2008). When we aggregate our data across foreign countries, the correlation between state level aggregate trade and state population is 93% for imports and 88% for exports respectively. We are therefore confident our trade dataset at the U.S. state  $\times$  foreign country level is not subject to severe measurement error.

17. Fifty-three countries have no FDI links with U.S. firms in our sample.

18. The historical information in this section is from Daniels (2002) and Thernstrom (1980). Also see Goldin (1994) for the political economy of U.S. immigration policy.

# **REVIEW OF ECONOMIC STUDIES**

TABLE 1
Summary Statistics

	All		Ancestry > 0	
		All	Bottom quintile	Top quintile
	(1)	(2)	(3)	(4)
Panel A: origin-destination pairs				
FDI dummy	0.018	0.031	0.003	0.127
American 2010 (in the second -)	(0.132)	(0.173)	(0.052)	(0.333)
Ancestry 2010 (in thousands)	(5.962)	(8,036)	(0.000)	2.852
Immigrants between 1990 and 2000 (in thousands)	0.023	0.042	0.000	0.199
iningrands between 1996 and 2000 (in diousands)	(1.070)	(1.443)	(0.001)	(3.221)
Immigrants between 2000 and 2010 (in thousands)	0.020	0.036	0.000	0.173
	(0.665)	(0.898)	(0.002)	(1.999)
Foreign-born 2010 (in thousands)	0.069	0.125	0.000	0.594
	(2.749)	(3.708)	(0.004)	(8.267)
Geographic distance (km)	9,122.393	8,397.379	9,142.553	7,463.619
Latitude difference (degree)	(5,802.105)	(5,705.716)	(4,299.372)	(2,980.255)
Lanuae anterence (degree)	(11 312)	(10.902)	(11 388)	(8 807)
No. of FDI relationships	0.196	0.351	0.028	1.620
F*	(5.486)	(7.396)	(1.461)	(16.294)
No. of subsidiaries in origin	0.033	0.060	0.003	0.270
	(1.345)	(1.813)	(0.281)	(3.844)
No. of parents in destination	0.015	0.027	0.001	0.123
	(0.399)	(0.537)	(0.103)	(1.175)
No. of workers employed at subsidiary in origin (in thousands)	0.039	0.069	0.010	0.319
	(4.941)	(6.661)	(1.298)	(14.750)
No. of subsidiaries in destination	0.068	0.122	0.011	0.562
No. of noncento in origin	(1.903)	(2.505)	(0.546)	(5.007)
No. of parents in origin	(2,282)	(3.077)	(0.580)	(6.811)
No. of workers employed at subsidiary in destination (in	0.050	0.088	0.027	0.392
thousands)	01020	0.000	0.027	0.072
,	(2.798)	(3.743)	(2.098)	(7.895)
Information demand index (standardized) <sup>a</sup>	0.599	0.735	0.197	1.489
	(1.000)	(1.092)	(0.443)	(1.448)
<u>N</u>	19,110	14,583	770	4,527
Panel B: countries				
Genetic distance	0.103	0.084	0.106	0.066
	(0.053)	(0.041)	(0.050)	(0.036)
N	155	119	18	25
Linguistic distance	0.950	0.937	0.990	0.920
A7	(0.110)	(0.121)	(0.010)	(0.114)
N Religious distance	132	103	8	20
Religious distance	(0.129)	(0.137)	(0.050)	(0.128)
Ν	131	101	8	25
Judicial quality	0.503	0.537	0.546	0.661
1	(0.208)	(0.214)	(0.224)	(0.202)
Ν	144	115	15	26
2010 Country diversity	0.442	0.405	0.433	0.239
	(0.269)	(0.256)	(0.246)	(0.197)
N	162	122	20	27
Panel C: counties				
2010 share of population with foreign ancestry	0.577	0.577	0.560	0.648
	(0.188)	(0.187)	(0.223)	(0.137)
2010 diversity of ancestries	0.790	0.789	0.764	0.838
	(0.075)	(0.075)	(0.071)	(0.077)
Ν	3,141	3,137	628	627

*Notes:* The table presents means (and standard deviations). Variables in Panel A refer to our sample of (country-county) pairs. Variables in Panel B refer to our sample of countries. Variables in Panel C refer to our sample of counties. Column 1 shows data for all observations. Columns 2 to 4 show all, the bottom quintile, and the top quintile of observations with positive ancestry, respectively. In Panel A, the FDI dummy is a dummy variable equal to 1 if the destination county has either subsidiaries or shareholders in the origin country. The details of variables in Panel B are given in the Supplementary Data Appendix. The ancestry-diversity variable is computed as 1 minus the Herfindahl index of ancestry group shares in each county. <sup>a</sup>The data are at the metropolitan area level.



Origins of Immigrants to the United States, pre-1880 to 2000. *Notes*: The left side depicts the share of total immigration to the United States in each census period for the largest five origin countries of US residents that claim foreign ancestry in the 2010 census: Germany, Britain, Ireland, Mexico, and Italy. The right side shows the the number of migrants (in millions) by continent of origin. See Section 1 of the main text and Supplementary Appendix A.1 for details

Until World War I, migration to the U.S. was largely unregulated. European migrants in particular faced few or no restrictions and came in large numbers. Figure 1 shows the extent and the changing composition of migration over time. Although the peak of British migration was passed before the beginning of our sample, the numbers for 1880 clearly show the effect of the potato famines and the subsequent large inflow of Irish migrants. The second big wave of migration in our sample is that of Germans in the aftermath of the failed revolutions of 1848 and the consolidation of the German empire under Prussian control in 1871. Similarly disrupted by political changes and an economic crisis in the South, Italians migrated to the U.S. in large numbers around 1910, followed by a peak in migrations from Eastern Europe and Russia in the years after the October Revolution. The inflow of migrants overall dropped dramatically during World War I, falling below 4 million during the period between 1910 and 1930.

While economic and political factors in the origin countries dominated the timing of these earlier European migrations, U.S. immigration policies became relatively more important during the 1920s. The first important step towards regulating the inflow of migrants was the Chinese Exclusion Act of 1882 that ended the migration of laborers, first from China, and then in following incarnations from almost all of Asia. These restrictions were followed by literacy and various other requirements that came into effect after 1917, culminating in the establishment of a quota system in 1921. The quota system limited the overall number of immigrants, reduced the flow of migrants from Southern and Eastern Europe, and effectively shut out Africans, Asians, and Arabs. Combined with the effects of the Great Depression, these new regulations led to negative net migration in the early 1930s and then a stabilization at relatively low levels of immigration. The quota system was abolished in 1965 in favour of a system based on skills and family relationships, leading both to a large increase in the total number of migrants and a shift in composition towards migrants from Asia and the Americas, in particular from Mexico.

Figure 2 maps the spatial settlement pattern of newly arrived immigrants in the U.S. over time. For each census from 1880 to 2010, we project the total number of new migrants from all origins to destination  $d, I_d^t$ , on destination and year fixed effects to account for general immigration time trends and persistent destination-specific effects. The figure shows the residuals from this projection, colour coded by decile. Migrants initially settled on the East Coast of the U.S. (in the mid-nineteenth century), and then the frontier for migrants moved to the Midwest (in the late-nineteenth century), to the West (1900–30), and to the South (in the 1980s). Starting in the



Destinations of Immigrants to the United States, pre-1880 to 2000. *Notes*: This figure maps immigration flows into US counties by census period. We regress the number of immigrants into US county d at time t,  $I_d^t$ , on destination county d and year t fixed effects, and calculate the residuals. The maps' color coding depicts the residuals' decile across counties and within census periods. Darker colours indicate a higher decile

1970s, we can also see graphically the increased settlement of migrants in urban centers, with a series of dark dots appearing around large urban areas.

In the next section, we use historical variation in both the timing of migrations from foreign countries to the U.S., and the timing of how attractive U.S. counties are for newly arriving migrants, as the basis of our identification strategy.

# 3. ANCESTRY AND FOREIGN DIRECT INVESTMENT

Before presenting formally our econometric model and results, we use a stylized historical example to describe the intuition behind our "leave-out push-pull" approach to identification.

The purpose of our identification strategy is to isolate variation in the distribution of presentday ancestry, which is independent of unobserved factors that could also affect the distribution of FDI.

Figure 3 illustrates our approach using two specific examples: that of migrations from Germany, with a migration peak in the pre-1900 period (corresponding to the failed 1848 revolution and the consolidation of the German empire under Prussian control), and that of Italy, with a peak in the 1900–30 period (triggered by the end of feudalism and demographic



NON-EUROPEAN IMMIGRATION: PRE-1900

German descendants in 2010



FIGURE 3

Migrants and Ancestors: Germany Pre-1900 and Italy 1900–1910. *Notes:* This figure contrasts Italian and German ancestry in 2010 (right panels), and non-European immigration patterns pre-1900 and 1910–1930 (left panel). The left two panels are created as in Figure 2, restricted to non-European immigration, and the periods pre-1900 and 1910–30.

The right two panels plot the county level residuals from a regression of log ancestry in 2010 on county, Italy and Germany fixed effects on the sample of European countries. The maps' colour coding depicts the residuals' decile in the distribution of residuals across counties. Darker colours indicate a higher decile

pressures, and ending with Mussolini's anti-emigration policies).<sup>19</sup> The top-left part shows the relative attractiveness of U.S. destinations for pre-1900 migrants, when German migrations to the U.S. peaked, as measured by the location choices of non-European migrants. At that time, most of these non-European migrants settled in the Midwest. Accordingly, we expect most German migrants from this initial wave to also have settled in the Midwest. The top-right part shows the distribution of U.S. residents with German ancestry in 2010, with disproportionately many in the Midwest. The bottom-left part shows the relative attractiveness of U.S. destinations for non-European migrants during the 1900–30 period, when Italian migrations to the U.S. peaked. At that time, the preferred destination for non-European migrants had shifted to the West and South. We expect many Italian migrants to have settled in these areas. The bottom-right part shows the distribution of Italian descendants in 2010, with relatively large populations in the West and South.

This is, in a nutshell, how we identify exogenous variation in the distribution of ancestry across U.S. counties. We use the interaction of time-series variation in the relative attractiveness of different destinations within the U.S. (measured by the destination choices of migrants from

19. In absolute terms, there were also large migrations from Italy prior to 1900, but arrivals from Germany and Ireland were far more numerous during that earlier period.

other continents) with the staggered arrival of migrants from different origins (measured by the number of migrants from that origin that migrated to other regions within the U.S.) to instrument for the present-day distribution of ancestry: if, coincidentally, large numbers emigrated from a given origin to the U.S. at the same time as a given location within the United States was attracting migrants from all over the world, then we expect large numbers of people in that U.S. location with ancestry from that foreign country.

Importantly, this interaction of "leave-out push-pull" factors is independent of plausible confounding factors that could make (or have made) a given destination in the U.S. more attractive for both migrations and FDI from a given origin country. Consider the example of Italian migrants in the 1900–30 period. Suppose many Italian migrants skilled at growing wine settled in U.S. regions favourable to wine growing (*e.g.* Napa county). The same unobserved factor (a climate favourable to wine-growing) may very well explain why there are both residents with Italian ancestry in Napa (descendants of wine makers), and FDI linkages between Napa and Italy (wine making multinationals), creating a spurious (not causal) correlation between ancestry and FDI. Our instruments remove this spurious correlation by predicting the number of Italians migrating to Napa 1900–30 using only the interaction of the share of non-European immigrants who settled in Napa with the number of Italians who settled outside the West Coast. Thus, if wine making ability were the only true driver of migrations from Italy (or other European countries) to Napa (or other counties on the West Coast), our leave-out push-pull instruments would predict zero Napa residents with Italian ancestry today.<sup>20</sup>

The same is true for migrations induced by reverse causality and most other confounding factors that might induce a spurious correlation between ancestry and FDI. For example, if the true cause of the large Italian presence in Napa was that an Italian car manufacturer randomly decided to invest in a Napa-based plant, and historically sent Italian workers to operate it, then this investment would affect realized migrations between Italy and Napa, but would again have no effect on the number of Napa residents with Italian ancestry predicted by our instruments.

For this approach to fail, a confounding factor that promotes migration and FDI would have to disproportionately cause large groups of migrants from two origins on two *different continents* to systematically migrate to the same destinations across at least two *different regions* in the *same census periods*. One example would be if migrants who are skilled at growing wine tended to emigrate from Algeria (a non-European country suitable for wine) towards Napa precisely at the same points in history when Italians also went to Napa and to the Champlain Valley in upstate New York (a non-West Coast region suitable for wine), and if Algerians in Napa were a large fraction of all migrants to Napa, and Italians in Champlain Valley a large fraction of all Italians arriving in the U.S.. We would then predict a large number of Napa residents with Italian ancestry, because of an unobserved factor (local climate) that directly affects both migrations and FDI from Italy and Algeria to Napa and Champlain Valley. We argue below that such occurrences are unlikely (Algerians did not in fact migrate to Napa in large numbers 1900–30), and offer a series of tests to gauge this possibility.

# 3.1. Identifying the causal impact of migrations

To formally evaluate the effect of the presence of descendants of migrants from a given origin on the probability that at least one firm within a given destination has an FDI link with a firm based

20. Incidentally, Figure 3 shows very few non-Europeans emigrated to Napa and Sonoma in 1900–30 (bottom left map), while many residents of Napa and Sonoma have Italian ancestry (bottom right map). Our identification strategy will *not* capture those descendants of Italian migrants in Napa and Sonoma. This is desirable, as both Italian ancestry and FDI linkages would likely be correlated with unobserved factors (local climate).

in the origin country (inward or outward), we estimate the structural gravity equation,<sup>21</sup>

$$\mathbf{1}[\mathrm{FDI}_{o,d} > 0] = \delta_o + \delta_d + \beta A_{o,d}^{2010} + X'_{o,d} \gamma + \varepsilon_{o,d}, \tag{1}$$

where  $\mathbf{1}[\text{FDI}_{o,d} > 0]$  is a dummy variable equal to 1 if any firm headquartered in destination *d* is either the parent or the subsidiary of any firm headquartered in origin *o* in 2014, zero otherwise.<sup>22</sup>  $A_{o,d}$  is a measure of common ancestry, usually calculated as the log of 1 plus the number of residents in *d* that report having ancestors in origin *o* in 2010, measured in thousands (we choose this functional form in anticipation of non-parametric results, but also show robustness to a wide range of alternative specifications—see Section 3.5).  $X'_{o,d}$  is a vector of control variables that always includes the geographic distance between *o* and *d*, and the difference in latitude between *o* and *d*.  $\delta_o$  and  $\delta_d$  represent a full set of origin and destination fixed effects, augmented in most of our specifications by fixed effects for the interaction between destination and continent of origin, and between origin and destination census region.<sup>23</sup> The error term  $\varepsilon_{o,d}$  captures all omitted influences, including any deviations from linearity.<sup>24</sup> Standard errors are clustered at the origincountry level, and our results are robust to alternative methods for calculating standard errors (see Section 3.5). The coefficient of interest is  $\beta$ , which measures the effect of ancestry on the probability that an FDI relationship exists between firms in *o* and *d*.

Equation (1) will consistently estimate the parameter of interest if  $\text{Cov}\left(A_{o,d}^{2010}, \varepsilon_{o,d}\right) = 0$ . As discussed above, this condition is unlikely to hold despite the inclusion of origin and destination fixed effects. First, origin-destination specific omitted factors might drive both economic transactions and migration flows, affecting both  $A_{o,d}$  and  $\mathbf{1}[\text{FDI}_{o,d} > 0]$ . (Skills and climate favourable to wine growing in our example above.) Second, past origin-destination specific migration flows might be the result of economic transactions such as FDI or trade, not their driver.<sup>25</sup> Third, ancestry might be selectively recalled because of past or present economic interactions. These challenges are not unique to our data, but are likely concerns with any data where ethnic linkages and economic transactions are simultaneously observed.

To address these concerns, we devise an IV strategy based on a simple dynamic model of migration: ancestry evolves recursively from the addition of new migrants to the existing stock of ancestry. We assume the combination of three forces determines the allocation of new migrants. A country-specific *push factor* drives migrants out of country o into the U.S.; a *pull factor* attracts migrants entering the U.S. to county d, irrespective of their origin; and a *recursive factor* corresponds to the tendency of newly arrived migrants to settle in communities where people with the same ancestry already live.

The following equation is a simple linear formulation of these assumptions.

$$A_{o,d}^{t} = a_{t} + a_{o,t} + a_{d,t} + b_{t}A_{o,d}^{t-1} + I_{o}^{t}\left(c_{t}\frac{I_{d}^{t}}{I^{t}} + d_{t}\frac{A_{o,d}^{t-1}}{A_{o}^{t-1}}\right) + v_{o,d}^{t}.$$
(2)

21. The gravity structure can be derived in a variety of models (Arkolakis *et al.*, 2012). See Carr *et al.* (2001), Razin *et al.* (2003), Head and Ries (2008), and Ramondo (2014) for applications to FDI.

22. We use this combined measure of inward and outward FDI because our main results are largely identical when separately considering inward and outward FDI. We report separate results for each direction below.

23. A census region is one of nine groupings of adjacent U.S. states listed in Supplementary Appendix Table 5.

24. We use a simple linear probability model, which allows for a straight-forward interpretation of the coefficient. As a robustness check, we also report results from a probit estimator; see footnote 29.

25. A real-life example of such reverse causality is the large Japanese ancestry in Scott County, Kentucky, which emerged after Toyota seconded Japanese workers to a newly built manufacturing facility in the 1980s.

The stock of residents of ancestry o in destination d at time  $t, A_{o,d}^t$ , depends on four terms. First, the constant terms  $a_t, a_{o,t}$ , and  $a_{d,t}$  control for residual forces, such as demographics, which may vary over time, space, and between different ethnic groups. Second, the term  $b_t A_{o,d}^{t-1}$  corresponds to the fact that ancestry is a stock variable that evolves cumulatively, where  $b_t$  modulates how ties to one's ancestry are passed from one generation to the next, including attenuation due to internal migrations. Third, the term  $I_o^t \left( c_t I_d^t / I^t + d_t A_{o,d}^{t-1} / A_o^{t-1} \right)$  is a linear interpretation of our assumption that net migrations are determined by the combination of push, pull and recursive factors. The push factor (the extent to which migrants are driven out of country o) is measured by the total number of migrants from country o entering the U.S. at time  $t_1 I_0^t$ . The pull factor (the degree to which county d is appealing to migrants at time t) is measured by the fraction of all migrants entering the U.S. who settle in county d from all origins,  $I_d^t/I^t$ . The recursive factor (the propensity of migrants to settle near their countrymen) is measured by the fraction of people with ancestry from o who already live in  $d_{o,d}^{t-1}/A_o^{t-1}$ . Intuitively, we assume that part of the allocation of new migrants from o across counties (the push factor  $I_0^t$ ) is proportional to the allocation of migrants from all countries (the pull factor  $I_d^I/I^t$ ), and part is proportional to the allocation of the existing stock of migrants from o (the recursive factor  $A_{od}^{t-1}/A_o^{t-1}$ ). The coefficients  $c_t$  and  $d_t$  control for the relative importance of the pull and recursive factors. If the pull factor is absent ( $c_t = 0$ ), our model collapses exactly to the Card (2001) model. Finally,  $v_{a,d}^t$ is a sequence of error terms that are potentially correlated with  $\varepsilon_{o,d}$ .

Equation (2) is not a suitable first stage because persistent forces are likely to shape both migrations and FDI, inducing a correlation between  $A_{o,d}^{t-1}$  and  $\varepsilon_{o,d}$ .<sup>26</sup> Therefore, an IV strategy following Card (2001), using variations in  $I_o^t$  and  $A_{o,d}^{t-1}$  as instruments, would not be suitable. We address this challenge by using the recursive structure of equation (2). Given that our data

We address this challenge by using the recursive structure of equation (2). Given that our data cover the vast majority of migration to the U.S. (more than 70 million immigrants, including the entire first and second generation of immigrants alive in 1880), we assume the initial condition  $A_{o,d}^{1880^{\circ}-1^{\circ}}=0, \forall (o,d)$  for simplicity. Solving (2) recursively, we get,

$$A_{o,d}^{2010} = \sum_{t=1880}^{2010} \left( a_t + a_{o,t} + a_{d,t} + c_t I_o^t \frac{I_d^t}{I^t} + v_{o,d}^t \right) \prod_{s=t+1}^{2010} (b_s + d_{o,s} I_o^s),$$
(3)

where the constant  $d_{o,s}$  only contains information on total migrations from o in previous periods.

This specification suggests plausibly exogenous variation in  $I_o^t(I_d^t/I^t)$  would allow the construction of an instrument for  $A_{o,d}^{2010}$ . By interacting a push factor,  $I_o^t$ , not specific to destination d but to all destinations in the U.S., and a pull factor,  $I_d^t/I^t$ , not specific to country o but to migrants from all countries, this formulation already rules out most plausible sources of endogeneity. However, our exclusion restriction could still be violated if  $I_{o,d}^t$ , or migrations from other origins similar to o, potentially correlated with  $\varepsilon_{o,d}$ , were a large fraction of  $I_o^t, I_d^t$  or  $I^t$ .

To address these concerns, we exclude migrants to d's census region from the push factor (we replace  $I_o^t$  by  $I_{o,-r(d)}^t$  in (3), where -r(d) means outside of d's census region), and migrants from o's continent from the pull factor (we replace  $I_d^t/I^t$  by  $I_{-c(o),d}^t/I_{-c(o)}^t$  in (3), where -c(o)

<sup>26.</sup> In the example above, a favourable climate for growing wine induces both migrations from Italy to Napa in 1900–30, a high  $A_{o,d}^{t-1}$  term, and many Italy-Napa wine making multinationals in 2014, resulting in a high  $\varepsilon_{o,d}$ .

means outside of o's continent). Our first-stage specification is thus

$$A_{o,d}^{2010} = \delta_o + \delta_d + \sum_{t=1880}^{2000} \alpha_t I_{o,-r(d)}^t \frac{I_{-c(o),d}^t}{I_{-c(o)}^t} + \sum_{n=1}^5 \delta_n P C_n + X_{o,d}' \gamma + \eta_{o,d}, \tag{4}$$

where  $\sum_{n=1}^{5} \delta_n P C_n$  stands for the first five principal components summarizing the information contained in the 758 higher-order terms  $I_{o,-r(d)}^s \cdots I_{o,-r(d)}^t I_{-c(o),d}^t / I_{-c(o),d}^t / V_{-c(o),d}^t / V_{-c(o),d}^t$ 

Our key identifying assumption is

$$\operatorname{Cov}\left(I_{o,-r(d)}^{t}\frac{I_{-c(o),d}^{t}}{I_{-c(o)}^{t}},\varepsilon_{o,d}|\operatorname{controls}\right)=0.$$
(5)

It requires that any confounding factors that make destination d more attractive for FDI from origin o do not simultaneously affect the interaction of the settlement of migrants from other continents with the total number of migrants from o settling in a different census region.

To further relax this assumption, most of our specifications also control for interactions of fixed effects that are symmetric to the construction of our instruments: the interaction between destination and continent-of-origin fixed effects  $(\delta_d \times \delta_{c(o)})$  and the interaction between origin and destination-census-region fixed effects  $(\delta_o \times \delta_{r(d)})$ .

These specifications are, by construction, robust to any confounding factors that operate within an origin-continent - destination-census-region pair. (In our example above, wine makers from Europe migrating to the West Coast.)

#### 3.2. The first-stage relationship

Table 2 shows our basic first-stage estimates of (4). Column 1 is the most parsimonious specification regressing our measure of ancestry on origin and destination fixed effects and the nine simple interaction terms  $\{I_{o,-r(d)}^{t}(I_{-c(o),d}^{t}/I_{-c(o)}^{t})\}_{t}$ . To interpret each coefficient as the marginal effect of migrations in a given period, without affecting the fit of the first stage, we sequentially orthogonalize each of the terms with respect to the previous interactions. For example, the coefficient marked  $I_{o,-r(d)}^{1910}(I_{-c(o),d}^{1910}/I_{-c(o)}^{1910})$  shows the effect of the residual obtained from a regression of  $I_{o,-r(d)}^{1910}(I_{-c(o),d}^{1910}/I_{-c(o)}^{1910})$  on the same interaction in 1880 and 1900.

All nine coefficients shown in column 1 are positive, and seven are statistically significant at the 1% level (Supplementary Appendix Figure 1 depicts the coefficients graphically; Supplementary Appendix Figure 4 depicts the fit of each coefficient). Even our earliest (pre-1880) snapshot of the cross-sectional variation in economic attractiveness to new migrants has left a significant imprint on the present-day ancestry composition of U.S. counties. The overall pattern of coefficients suggests a hump-shape, where very recent waves of migrants have a smaller impact on current ancestry than migrations a few decades back, but the effect of past migrations eventually fades after about one century. An exception to the general pattern is the smaller and insignificant coefficient for 1920–30. A likely explanation is the Great Depression, which induced large reverse migrations

<sup>27.</sup> Principal component analysis (eigenvalue decomposition) is simply a means for compactly summarizing the variation contained in the 758 higher-order terms. In our standard specification, the first five components summarize 99.99% of the variation. To the extent that the higher order terms are valid instruments, the first five principal components are valid instruments as well. Our results are robust to adding these terms or not.

			2	, Log and	estry 2010				Ancestry 2010
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)
$I_{o,-r(d)}^{1880} \times \frac{I_{-c(o),d}^{1880}}{I^{1880}}$	$0.057^{***}$	0.056***	0.045***	0.035***	0.056***	0.056***	0.056***		2.145***
-c(0)	(0.013)	(0.012)	(0.011)	(600.0)	(0.007)	(0.007)	(0.008)		(0.290)
$I_{o,-r(d)}^{1900}  imes rac{I_{-c(o),d}^{1900}}{I^{1900}}$	0.097***	0.095***	0.067**	0.068***	0.099***	0.099***	$0.100^{***}$	0.106**	3.634**
( <i>a</i> )	(0.032)	(0.031)	(0.028)	(0.019)	(0.033)	(0.033)	(0.036)	(0.043)	(1.425)
$I_{o,-r(d)}^{1910}  imes rac{I_{-c(o),d}^{1910}}{I^{1910}}$	$0.192^{***}$	0.193***	0.145***	0.123***	$0.137^{***}$	$0.137^{***}$	$0.132^{**}$	$0.123^{**}$	5.646**
(0)0	(0.039)	(0.039)	(0.042)	(0.031)	(0.050)	(0.050)	(0.054)	(0.049)	(2.345)
$I_{0,-r(d)}^{1920}  imes rac{T_{-c(0),d}^{72,0}}{T^{1920}}$	0.205***	$0.209^{***}$	$0.176^{***}$	$0.174^{***}$	$0.283^{***}$	$0.283^{***}$	$0.249^{***}$	$0.276^{***}$	$14.726^{***}$
( <i>a</i> )0	(0.070)	(0.070)	(0.061)	(0.052)	(0.045)	(0.045)	(0.047)	(0.041)	(3.012)
$I_{o,-r(d)}^{1930}  imes rac{I_{royd}^{1930}}{I^{1930}}$	0.062	0.061	0.061	0.035	0.079	0.079	$0.065^{*}$	0.078	$11.812^{***}$
	(0.056)	(0.056)	(0.056)	(0.048)	(0.051)	(0.051)	(0.034)	(0.051)	(2.855)
$I_{0,-r(d)}^{1970}  imes rac{T_{-c(0),d}^{1970}}{T^{1970}}$	$0.183^{***}$	$0.184^{***}$	$0.163^{***}$	$0.149^{***}$	$0.149^{***}$	$0.148^{***}$	$0.150^{***}$	$0.151^{***}$	6.256***
-0(0)	(0.038)	(0.038)	(0.036)	(0.031)	(0.028)	(0.029)	(0.027)	(0.029)	(0.669)
$I_{o,-r(d)}^{1980}  imes rac{T_{row,d}^{1980}}{T^{1980}}$	$0.173^{***}$	$0.173^{***}$	$0.174^{***}$	$0.169^{***}$	$0.214^{***}$	$0.214^{***}$	$0.205^{**}$	$0.213^{***}$	$18.694^{***}$
	(0.066)	(0.066)	(0.064)	(0.061)	(0.076)	(0.077)	(0.080)	(0.075)	(2.390)
$I_{0,-r(d)}^{1990} \times \frac{I_{1090}^{1200}}{I_{1090}^{1000}}$	$0.123^{***}$	$0.124^{***}$	$0.124^{***}$	$0.111^{**}$	$0.101^{**}$	$0.101^{**}$	$0.115^{**}$	$0.102^{**}$	$10.786^{***}$
-0.03	(0.048)	(0.048)	(0.048)	(0.044)	(0.045)	(0.045)	(0.048)	(0.044)	(3.675)
$I_{0,-r(d)}^{2000}  imes rac{I_{-c(0),d}^{2000}}{I^{2000}}$	0.020	0.019	0.026	$0.026^{*}$	$0.046^{***}$	$0.046^{***}$	$0.039^{**}$	$0.046^{***}$	$5.194^{***}$
	(0.017)	(0.017)	(0.016)	(0.015)	(0.017)	(0.017)	(0.016)	(0.017)	(1.148)
$I_{o,-r(d)}^{2010}  imes rac{I_{c(0),d}^{2010}}{I_{2010}^{2010}}$							$0.317^{***}$		
600-							(0.089)		
Kleibergen Wald rk statistic	10.608 20.52	10.958 20.52	8.327	9.607	162.194	158.125	195.423	142.800	910.331
Stock-10g0 3% clitical values Stock-Yogo 10% critical values	11.46	11.46	11.46	20.02 11.46	21.10 11.52	21.10 11.52	11.51	21.10 11.52	21.10 11.52
$R^2$	0.56	0.56	0.66	0.72	0.73	0.73	0.73	0.73	0.49
N	612,495	612,495	612,495	612,495	612,495	612,495	612,495	612,495	612,495
Destination FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Distance	No	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Latitude difference	No	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Destination × continent FE	No	No	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Origin × census region FE	No No	No	No No	Yes	Yes	Yes	Yes	Yes	Yes
Principal components	No	NO	No	No 2	Yes	Yes	Yes	Yes	Yes
3rd order poly in dist and lat	No	No	No	No	No	Yes	No	No	No
<i>Notex:</i> The table presents coefficient t tered at the origin country level. In col 2010). In column 9 the dependent vari where <i>I</i> stands for the number of	sstimates of our first lumns 1–8 the deper iable is the level of a of miorants from 0.0	stage equation (4) al dent variable is the 1 ncestry in 2010 (agai who settle in destinal	i the country-county og of 1 plus the numb in in thousands). The ions not in the same	level. All specificat ber of residents of th excluded instrumer	ions control for orig ie county in 2010 th ats are, for each cens in neriod t and It	in and destination fixe the treport having ancest us period, interaction $\sqrt{t}$ for the fract	deffects. Standard e ors in the origin cou s of pull and pushfac	rrors are given in pare ntry, measured in tho tors in migration, $I_{o,-}^{t}$	antheses and are clus- last (Log Ancestry $r(d) \begin{pmatrix} I^{t} \\ -c(o), d \end{pmatrix} \begin{pmatrix} I^{t} \\ -c(o) \end{pmatrix}$ ),
as $o$ who settle in county $d$ . Columns $\hat{z}$	3-9 also include the	first five principal co	information of higher-	order interactions of	of these factors. $*, *$	$\frac{1}{c} \frac{d}{dr} I - \frac{1}{c(o)}$ 101 up the stati	stical significance at	tthe 10%, 5%, and 19	6 levels, respectively.

. • ALS A TABLE 2  $ER_{c}$ T. 16

# **REVIEW OF ECONOMIC STUDIES**

of recently arrived migrants, demonstrating our model is less well suited for periods with negative net migration.

Taken together, the nine simple interactions incrementally increase the  $R^2$  of the regression by 4 percentage points and explain about 9% of the variation in ancestry not explained by origin and destination fixed effects. Column 2 controls for distance and latitude difference. Columns 3 and 4 add destination × continent-of-origin and origin × destination-census-region fixed effects, respectively. Columns 1–4 estimate (4) under the restriction that the recursive factor is irrelevant ( $d_t = 0$  in (2)). We relax this in column 5 and add the principal components of the higher order interaction terms. Column 6 includes third-order polynomials in the distance and latitude difference between o and d. Column 7 includes migration data from 2005 to 2010 ACS survey. Column 8 drops migration prior to 1880. Column 9 estimates (4) in levels rather than logs.

Our standard specification is in column 5. The Kleibergen–Papp Wald rk-statistic against the null of weak identification is 162.2, well above the Stock and Yogo critical values.<sup>28</sup> We reject the null that our instruments are jointly irrelevant in the first stage across all specifications.

#### 3.3. IV results

In our IV estimation, we explicitly test the hypothesis that an increase in the number of descendants from a given origin increases the probability that at least one local firm engages in FDI with that country. The dependent variable is a dummy equal to one if either a parent foreign firm from origin country o owns a U.S. subsidiary in destination U.S. county d (inward FDI), or if a U.S. parent in d owns a foreign subsidiary in o (outward FDI). The results are in Table 3.

In column 1 we estimate equation (1) while instrumenting (the log of) ancestry in 2010 with the simple interaction terms  $\{I_{o,-r(d)}^t(I_{-c(o),d}^t/I_{-c(o)}^t)\}_t$  and controlling for origin and destination fixed effects, distance, and latitude difference. The coefficient estimate on ancestry is 0.231 (SE = 0.023), statistically significant at the 1% level. Supplementary Appendix Figure 2 shows the corresponding reduced form results graphically. All nine coefficients are greater than zero, and seven of them are statistically significant at the 5% level. Destinations that received an (exogenous) increase in the number of migrants from a given origin in any of the nine consecutive waves of immigration thus tend to have a significantly higher probability of engaging in FDI with that origin today. The coefficient of interest falls slightly to 0.190 (SE = 0.024) in column 2 when we add the first five principal components of the higher-order interactions to our set of instruments. Column 3 shows our standard specification. The estimate, 0.187 (SE = 0.024), implies that doubling the number of residents with ancestry from a given origin relative to the sample mean (from 316 to 632) increases by 4 percentage points the probability that at least one firm engages in FDI with that origin.<sup>29</sup> This specification includes destination  $\times$  continent-of-origin fixed effects and origin × destination-census-region fixed effects. Reassuringly, adding these 17,460 fixed effects has almost no effect on our coefficient of interest (0.187, SE = 0.024 versus 0.190, SE = 0.024).

Comparing this estimate with the same column in panel B shows that it is about 25% larger than the corresponding OLS coefficient. The endogenous assignment of migrants within the U.S. thus appears to induce a downward bias in the OLS coefficient, consistent with a simple extension

29. Using  $\hat{\beta} = 0.187$  from column 3 in Table 3 in equation (1), we have:  $1[\text{FDI}_{o,d} > 0|\text{Ancestry}_{o,d} = 632] - 1[\text{FDI}_{o,d} > 0|\text{Ancestry}_{o,d} = 316] = 0.187 \left( \ln \left( 1 + \frac{632}{1000} \right) - \ln \left( 1 + \frac{316}{1000} \right) \right) \approx 0.0402$ . An IV probit estimate of the same specification yields a marginal effect of Log Ancestry 2010 on Pr[FDI > 0] of 0.104 (SE = 0.037).

<sup>28.</sup> The Hansen J test statistic is 15.891 with a *p*-value of 0.255. We thus fail to reject the null that our instruments are uncorrelated with the error term and correctly excluded from the second-stage regression.

Panel A: IV			FD	I 2014 (Dum	my)		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Log ancestry 2010	0.231***	0.190***	0.187***	0.187***	0.189***	0.198***	0.191***
	(0.023)	(0.024)	(0.024)	(0.024)	(0.030)	(0.023)	(0.024)
Log distance	0.007	0.004	0.024	0.009	0.030	0.026	-0.027
-	(0.010)	(0.009)	(0.029)	(0.033)	(0.040)	(0.030)	(0.027)
Ν	612,495	612,495	612,495	612,495	459,150	612,495	612,300
Panel B: OLS			FD	I 2014 (Dum	my)		
Log ancestry 2010	0.173***	0.173***	0.149***	0.149***	0.145***	0.149***	0.161***
	(0.016)	(0.016)	(0.018)	(0.018)	(0.019)	(0.018)	(0.019)
$R^2$	0.2967	0.2967	0.3635	0.3635	0.3920	0.3635	0.3930
Ν	612,495	612,495	612,495	612,495	459,150	612,495	612,495
Destination FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Origin FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Principal components	No	Yes	Yes	Yes	Yes	Yes	Yes
Destination × continent FE	No	No	Yes	Yes	Yes	Yes	Yes
Origin $\times$ census region FE	No	No	Yes	Yes	Yes	Yes	Yes
3rd order poly in dist and lat	No	No	No	Yes	Yes	No	No
Agricultural similarity (cosine)	No	No	No	No	Yes	No	No
$I_{a}^{2010} = r(d) (I_{-c(a)}^{2010} / I_{-c(a)}^{2010})$	No	No	No	No	No	Yes	No
Origin $\times$ state FE	No	No	No	No	No	No	Yes

 TABLE 3

 Second-stage: The effect of Ancestry on FDI

*Notes:* The table presents coefficient estimates from IV (Panel A) and OLS (Panel B) regressions of equation (1) at the country-county level. The dependent variable in all panels is a dummy indicating an FDI relationship between origin *o* and destination *d* in 2014. The main variable of interest is *Log Ancestry 2010*, instrumented using various specifications of equation (4). In all columns in Panel A, we include  $\{I_{o,-r(d)}^{t}(I_{-c(o)}^{t})\}_{t=1880,...,2000}$  as excluded instruments. Columns 3–7 also include the first five principal components of the higher-order interactions of push and pull factors as instruments. To avoid re-calculating the principal components across specifications, we include the 2010 wave in their calculation for all columns; however, the inclusion of the 2010 wave has essentially no effect on our results. For example, the standard specification in column 3 without including the 2010 wave in the calculation of principal components yields 0.187 (0.024). Column 5 also includes the interaction of the push and pull factor constructed using data from the 2006 to 2010 American Community Survey. All specifications control for log distance, latitude difference, origin, and destination fixed effects. Standard errors are given in parentheses. Standard errors are clustered at the origin country level. \*, \*\*, and \*\*\* denote statistical significance at the 10%, 5%, and 1% levels, respectively. (We also run an IV probit regression using the specification in column 2 yielding a marginal effect evaluated at the mean of *Log Ancestry 2010* on FDI equal to 0.104\*\*\*(0.037).)

of the Heckscher–Ohlin model: migrations tend to be driven by differences in factor endowments (creating differences in wages between origin country and destination county), while FDI flows are driven by similarities in factor endowments (as firms use FDI to export their technology to countries with a similar mix of factor endowments).<sup>30</sup>

Another useful way to gauge the relative importance of ancestry is its partial  $R^2$  relative to the other controls. Taken together, the standard gravity terms, that is, the origin and destination fixed effects, distance, and latitude difference, explain 20.3% of the variation in the FDI Dummy. Adding ancestry to these variables in a simple OLS specification (panel B) raises the  $R^2$  by 9 percentage points, half as much variation as the combined explanatory power of the economic fundamentals reflected in the gravity terms (although this effect is not necessarily causal).<sup>31</sup>

30. Naturally, the OLS coefficient may also be biased downward simply because ancestry is measured with error.

31. Instead adding our nine simple interactions to the standard gravity terms, thus running the most parsimonious reduced form, raises the  $R^2$  by 1.5 percentage points, and adding them in combination with the five principal components by 2 percentage points. These numbers are a lower bound on the importance of common ancestry for FDI, since it only accounts for the part of the causal effect of ancestry which is picked up by our instruments.

The remaining columns of Table 3 probe the robustness of this result. Column 4 adds a thirddegree polynomial in distance and latitude difference to capture non-linear effects of distance; column 5 adds an interaction term for the contemporaneous 2010 migrations in the first stage (as in column 7 of Table 2); and column 6 adds a more stringent set of origin × destination-state fixed effects, exploiting only variation within U.S. states. The coefficient estimate remains remarkably stable and highly statistically significant across specifications.

#### 3.4. The communist natural experiment and alternative instruments

The main potential challenge to these results is that, despite our best efforts, confounding factors that make a destination more attractive for both migration and FDI from a given origin may still, in some complicated way, be correlated with our instruments, although they only use information about migrations from other continents and to other census regions. In this section, we address this challenge using a natural experiment and a set of alternative instrumentation strategies.

**3.4.1. Communist natural experiment.** We begin by combining our IV with a natural experiment that allows us to focus on *changes* in FDI and *changes* in ancestry: the periods of economic isolation between the U.S. and communist countries during parts of the twentieth century. These periods are 1918–90 for the Soviet Union, 1945–80 for China, 1975–96 for Vietnam, and 1945–89 for Eastern Europe (the non-Soviet members of the Warsaw pact). They provide a useful experiment for two reasons. First, we can confidently assume the prospect of FDI, outlawed for political reasons, did not drive migrations during those periods (ruling out reverse causality). Second, the specification is similar to a difference-in-difference, measuring how cross-sectional variations in ancestry driven only by the inflow of migrants over a period of exclusion explain changes in FDI, from zero during the exclusion period to its current level in 2014.

Table 4 shows estimates of (1) for each of these countries or sets of countries, using as instruments only migration waves that occurred during the period of isolation. For each country, we find a large and significant causal impact of ancestry on FDI that is remarkably similar to our full-sample estimates from Table 3 (*e.g.* 0.185 (SE = 0.019) for the Soviet Union alone versus 0.187 (SE = 0.024) in our standard specification from Table 3). An exception to this rule is Vietnam, which shows a coefficient less than half the size of the other cases (0.089, SE = 0.036). Plausibly, this lower coefficient reflects the fact that the majority of Vietnamese immigrants who arrived during the 1975–96 period were granted entry for aiding the U.S. cause during the Vietnam war, and the communist government who defeated them is still in power in 2014 and controls FDI. Vietnamese Americans might thus plausibly be in a worse position than descendants of migrants from other countries to generate FDI between the U.S. and their ancestral country. Pooling across all former Communist countries, we again find a coefficient very close to that of our standard specification in Table 3 (0.206, SE = 0.031). These results strongly suggest reverse causality does not drive our results, and our exclusion restriction is likely valid.

**3.4.2.** Alternative instruments. The main remaining challenge to our approach is that a common unobserved characteristic of destinations in two different census regions may still directly affect FDI, while also disproportionately causing large groups of migrants from two origins on two different continents to systematically migrate at the same time to the same destinations across multiple census regions (*e.g.* the Italy–Algeria and Napa-Champlain Valley example above).

				_	
			FDI 201	4 (Dummy)	
	(1)	(2)	(3)	(4)	(5)
Log Ancestry 2010	0.185*** (0.019)	0.237*** (0.024)	0.089** (0.036)	0.172*** (0.021)	0.209*** (0.032)
Ν	3,141	3,141	3,141	18,846	28,269
Destination FE Origin FE	No No	No No	No No	No Yes	Yes Yes
Countries considered Period of ec. isolation Instruments from	Soviet Union 1918–90 1920–90	China 1945–80 1970–80	Vietnam 1975–96 1980–90	Eastern Europe 1945–89 1970–80	All communist countries

 TABLE 4

 The Effect of Ancestry on FDI: The Communist Natural Experiment

*Notes:* The table presents coefficient estimates from IV regressions of equation (1) at the country-county level. The estimates correspond to column 1 of Table 3 except in the inclusion of fixed effects, which is specified. Each column uses data from a subset of origin countries: Soviet Union (column 1), China (column 2), Vietnam (column 3), as well as Albania, Bulgaria, Czechoslovakia, Hungary, Poland, and Romania (column 4). The dependent variable in all columns is a dummy indicating an FDI relationship between origin country o and destination county d in 2014. In all specifications the instruments are constructed as in column 3 of Table 3; however, we only include as instruments interaction terms containing measures of pull and push factors in migrations that occur during the years of economic isolation from the United States indicated above. The remaining variables are included as controls. All specifications control for log distance, latitude difference, and origin fixed effects. Standard errors are given in parentheses and are robust. \*, \*\*, and \*\*\* denote statistical significance at the 10%, 5%, and 1% levels, respectively.

If such a confounding factor were driving our results, we would expect that excluding from the construction of our instruments countries with either correlated migrations, or correlated ancestry, would have a large effect on the coefficient of interest. We show below they do not.

The first specification in Panel A of Table 5 excludes from the construction of our pull factors foreign countries that tended to send migrants towards the U.S. at the same time as a given origin country. For each pair of countries, we compute the correlation coefficent over time of aggregate emigration to the U.S., and exclude from the (o, d, t) pull factor all migrations to d at t from origin country o' if the correlation with o is above 0.5 and significant at the 5% level. Panel A of Supplementary Appendix Figure 5 shows these correlations graphically: while correlations within continents (along the diagonal) tend to be larger, there is also a strong correlation in the timing of migrations between some African, American, and Asian countries.<sup>32</sup> Using this alternative approach, we exclude on average sixty-five countries, of which on average eighteen are in the same continent (our baseline leave-out category).

The second specification instead excludes from our instruments foreign countries which have a similar distribution of ancestry within the U.S. in 2010. For each country pair, we compute the correlation across destination counties of the number of residents with a given ancestry, and again exclude origin countries if the correlation coefficient is above 0.5 and significant at the 5% level. Panel B shows these correlations graphically, with a strong correlation in the distribution of ancestry between Asian and European countries. Using this approach, we exclude on average thirty-six countries, of which on average thirteen are in the same continent.

Panel A of Table 5 shows that our coefficient estimates vary little with these alternative sets of instruments: 0.181 (SE = 0.027) when excluding countries with correlated migrations over time; and 0.217 (SE = 0.030) when excluding countries with correlated ancestry over

<sup>32.</sup> Most, if not all, forces that induce such correlations in practice are fully innocuous for our identification. For example, changes in U.S. immigration laws or innovations in transportation technologies might simultaneously affect the push factors of several countries. In this sense, there is no general reason to exclude certain countries with high correlation from the construction of our instruments.

#### BURCHARDI ET AL.

#### MIGRANTS, ANCESTORS, AND FOREIGN INVESTMENTS21

Panel A: variations of leave-out categories	FDI 2014 (Dummy)
Excluding origins with correlated migration flows: $I_{a,-r(d)}^{t} \times (I_{-s^{1}(a),d}^{t}/I_{-s^{1}(a)}^{t})$	0.181***
(-) $(-)$ $(-)$ $(-)$ $(-)$ $(-)$	(0.027)
Excluding origins with correlated 2010 ancestry stock: $I_{o,-r(d)}^{t} \times (I_{-s^{2}(o),d}^{t}/I_{-s^{2}(o)}^{t})$	0.217***
	(0.030)
Excluding states adjacent to the destination: $I_{a - adi(d)}^{t} \times (I_{-c(a)}^{t} d / I_{-c(a)}^{t})$	0.192***
$\sigma_{i}$ $\omega_{i}(\omega)$ $\sigma_{i}(\sigma_{i})$ $\sigma_{i}(\sigma_{i})$	(0.022)
Panel B: using subsets of instruments for identification	FDI 2014 (Dummy)
Only migrations 1880–1930	0.209***
	(0.037)
Only migrations 1970–2000	0.175***
	(0.021)
Only migrations 1900–2000	0.187***
	(0.024)

TAB	LE 5
Alternative	instruments

*Notes:* This table presents coefficient estimates from instrumental variable regressions that are variations of our standard specification (column 3 of Table 3). Each row lists the coefficient estimate on *Log Ancestry 2010*. Panel A shows alternative specifications of our leave-out instrument: we exclude from the pull factor all countries for which the time correlation of total migration to the US with o's migration to the US is greater than 0.5 and significant at the 5% level; we exclude from the pull factor all countries for which the correlation of 2010 ancestry across the US with o's ancestry across the US is greater than 0.5 and significant at the 5% level; we exclude from the pull factor all countries for which the correlation of 2010 ancestry across the US with o's ancestry across the US is greater than 0.5 and significant at the 5% level; and in the push factor of o we exclude migrations to any state adjacent to the state of d including the state itself, respectively. In Panel B, we use throughout the interacted instrument of our standard specification but each specification in this panel uses as instruments only the simple interaction terms from a subset of the full time period covered by our data. Standard errors are given in parenthesis and clustered at the country level. \*, \*\*, and \*\*\* denote statistical significance at the 10%, 5%, and 1% levels, respectively.

space. The coefficient also remains stable when we exclude migrations to all adjacent states,  $I_{o,-adj(d)}^t(I_{-c(o),d}^t/I_{-c(o)}^t)$ , rather than the surrounding census region (0.192, SE = 0.022). Supplementary Appendix Tables 6 and 7 show that the same is true across a wide range of similar variations in our leave-out categories: our coefficient estimates remain stable across all variations that exclude at least the own origin-destination pair. We conclude that our original approach (excluding the own continent-of-origin and destination-census-region) is valid and that no spurious correlations are driving our results.

Panel B of Table 5 presents results using subsets of our instruments. The first row uses as instruments only the simple interactions from the first half of the time period covered by our migration data (1880–1930), the second row only from the second half (1970–2010), while the third row excludes migrations from the first census (1880) from the set of our instruments, as these might be more related to stocks than flows. The coefficient of interest again remains stable at 0.209 (SE = 0.037), 0.175 (SE = 0.021), and 0.187 (SE = 0.024) respectively.<sup>33</sup> Supplementary Appendix Table 8 replicates our results using data on FDI from 2007 rather than 2014 and data on ancestry from 2000 rather than 2010, again with little effect on our results. We conclude that our results are not driven by specific vintages of migrations, pre-, post-WWII, or other.

#### 3.5. Additional robustness checks and intensive margin

**3.5.1. Functional form.** In our main specification, we measure our ancestry variable as the log of one plus the number of residents with foreign ancestry, measured in thousands,  $A_{o,d}^{2010} = \ln\left(1 + \frac{1}{1000}\operatorname{Ancestry}_{o,d}^{2010}\right)$ . In Supplementary Appendix Table 9, we offer a formal test

33. The first two coefficients are not statistically different; the *p*-value for a test of equality is 0.202.

to justify this choice of functional form by performing a non-linear least squares estimation of

$$\mathbf{1}[\mathrm{FDI}_{o,d} > 0] = \delta_o + \delta_d + \beta \ln \left( 1 + \pi \operatorname{Ancestry}_{o,d}^{2010} \right) + X'_{o,d}\gamma + \varepsilon_{o,d}, \tag{6}$$

again including the same covariates as in our simple specification from column 2 in Table 3. We find a point estimate of  $\beta = 0.1683$  and  $\pi = 0.0010$ . This finding forms the basis for our choice of functional form applied throughout the article. This functional form is convenient because it offers a compact way to model the non-linear impact of ancestry. For small ancestry (Ancestry<sub>o,d</sub>  $\ll$  1000), the function  $\ln(1 + \text{Ancestry}_{o,t}/1000)$  is approximately linear in  $\text{Ancestry}_{o,d}$ . For large ancestry (Ancestry<sub>o,d</sub>  $\gg$  1000), it behaves approximatively like  $\ln(\text{Ancestry}_{o,d})$ . So for a small number of residents with foreign ancestry o, (1) means that increasing ancestry by 1,000 increases the probability of at least one FDI link to o by  $\beta$  percentage points; for a large number of residents with ancestry by 1% increases the probability of at least one FDI link so  $\beta$  percentage points are stry on FDI link by  $\beta$  percentage points. Supplementary Appendix Figure 6 plots the average number of FDI links across centiles of the distribution of ancestry, and shows the effect of ancestry on FDI is highly concave.

In Supplementary Appendix Table 10, we further sow the robustness of our results to using a range of alternative functional forms, and to using measures of ancestry from the, 1980, 1990 and 2000 censuses, instead of 2010. Supplementary Appendix Table 11 shows our main results are robust to varying the cutoff for ownership at which we consider a foreign firm to be a subsidiary or parent (from 5% to 50%). Supplementary Appendix Table 8 shows our results are robust to using FDI in 2007 (instead of 2014).

**3.5.2. Standard errors.** Supplementary Appendix Table 12 shows our standard specification from column 3 of Table 3 using alternative standard errors. It reports robust standard errors; standard errors clustered by origin, destination, state, continent, and state-country cells. Among all these simple analytic standard errors, clustering by origin, as we do throughout the article, is the most conservative choice. Doing so allows for arbitrary correlation in the error term across multiple destinations for a given origin, including for spatial correlation of errors. The specification we use throughout the article thus allows for more flexible patterns of spatial correlation than for example the standard error correction as proposed by Conley (1999).

A possible concern is that errors may still be correlated across origin countries. However, standard errors designed to adjust for such correlations (clustering by county or state) are narrower, suggesting that any such patterns in the error structure are — if they were present — absorbed by the rich set of fixed effects and controls contained in our standard specification. Consistent with this view, the table also shows that standard errors double clustered at county-plus-country and state-plus-country level, as well as various block-bootstrapped standard errors are either narrower or only very marginally wider than those in our standard specification. The conclusion that our results are robust to alternative standard error specifications also carries over to the other main results of our paper (see Supplementary Appendix Table 13).

An alternative approach to detecting any tendency to over-reject the null is to reassign the "treatment" to a different set of outcome observations, in the spirit of Fisher's randomization inference procedure. We assign the interaction between push and pull factors for country o to randomly selected other countries and calculate the *t*-statistic on the coefficient of interest. Reassuringly, across 1000 random assignments, the *t*-statistic rejects the null of no treatment effect in favour of the alternative of a positive treatment effect in only 2.7% of the cases.





Heterogeneous Effects across Countries and Counties. *Notes:* This figure shows funnel plots of the estimated coefficients and standard errors from separate IV regressions of the FDI dummy on Log 2010 Ancestry for each origin country (left) and destination US counties (right). In all regressions, we use  $\{I_{o,-r(d)}^t(I_{-c(o),d}^t/I_{-c(o)}^t)\}_{t=1880..2000}$  and principal components as excluded instruments, and control for log distance as well as latitude difference. *x*-axis: estimated coefficients. *y*-axis: reciprocal of estimated standard errors on ancestry. Circle sizes are proportional to country ancestry (left) and county population (right). Circles above the  $y=\pm 1.96/x$  curve indicate statistically significant coefficients. See Section 3.5 for details

**3.5.3. Placebo tests.** To assess whether our instrument reliably isolates push factors specific to only one country, or is correlated with omitted variables that affect FDI with other countries, Supplementary Appendix Table 14 assigns the interaction between push and pull factors for a given origin to a quasi-randomly selected other country: its nearest neighbour in alphabetic order (panel A), or its nearest neighbour in alphabetic order in a *different* continent (panel B). The coefficient estimates are always near zero and statistically insignificant, suggesting our instrument is not picking up any artificial correlation (positive or negative) between the push factors for different countries.

**3.5.4.** Robustness in sub-samples and heterogeneous effects. Figure 4 shows results from separate regressions for all 112 origin countries (left panel) and the 100 largest U.S. counties (right panel).<sup>34</sup> Each figure is a funnel plot of the country/country-specific coefficients on ancestry against the reciprocal of their standard errors, where the circles reflect the relative shares in ancestry and U.S. population, respectively. The coefficients are significant at the 5% level for 84 out of 112 countries and 99 out of 100 counties. This further demonstrates that our results are not driven by any specific subsample or outliers.

**3.5.5.** Inward and outward FDI. We estimate our standard specification from column 3 of Table 3 separately for inward FDI, a dummy equal to 1 if at least one firm in U.S. county d is a subsidiary of a parent in foreign country o, and for outward FDI, a dummy equal to 1 if at least one firm in U.S. county d is the parent of a subsidiary in foreign country o. The coefficients for both outward and inward FDI are positive, statistically significant, and close to our baseline

<sup>34.</sup> Supplementary Appendix Figure 7 shows a similar figure for six individual sector groups. Supplementary Appendix Table 15 shows results from separate regressions for the five largest origins (by number of descendants, panel A) and destinations (in total number of foreign ancestry, panel B). Supplementary Appendix Tables 16, 17, and 18 show the results from separate regressions for all countries, all sectors, six sector groups, and small and large firms.



FIGURE 5

Ancestry and FDI: Germany and Britain; Los Angeles and Cook counties. *Notes:* The figure shows conditional scatterplots from regressions corresponding to column 3 of Supplementary Appendix Table 19, restricted to one origin country (top parts: Germany and Britain) or one destination county (bottom parts: LA and Cook counties). The solid line depicts the fitted regression line, controlling for distance and latitude difference. *x*-axis: predicted log 2010 ancestry. *y*-axis: log # of subsidiaries in Germany (top left) and Britain (to right) for firms in each US county, log # of subsidiaries of LA (bottom left) or Cook county (bottom right) based firms in each origin country

estimates. We find a somewhat stronger impact of ancestry on outward FDI,  $\beta_{out} \approx 0.2$ , than on inward FDI,  $\beta_{in} \approx 0.15$ , although the coefficients are not statistically different.

**3.5.6.** The intensive margin of FDI. Figure 5 shows ancestry has a positive and significant impact on the intensive margin of FDI (the number of foreign subsidiaries), with no obvious outliers, for the largest countries and counties in our sample: Germany and Britain (top parts), and LA and Cook counties (bottom parts). Supplementary Appendix Table 19 systematically estimates the impact of ancestry on the intensive margin of FDI,

$$\ln \operatorname{FDI}_{o,d} = \delta_o + \delta_d + \kappa A_{o,d}^{2010} + X'_{o,d} \gamma + \zeta_{o,d},$$
(7)

where  $FDI_{o,d}$  corresponds to various measures of the volume of FDI between *o* and *d* and where we instrument  $A_{o,d}^{2010}$  with the same first-stage equation (4) as earlier.<sup>35</sup> We use various measures

35. To correct the selection bias from dropping log(zero)'s, we implement a Heckman (1979) correction, as in Helpman *et al.* (2008). We first estimate an IV probit regression at the extensive margin,

$$\rho_{o,d} = \Pr\left(\text{FDI}_{o,d} > 0 | \text{observables}\right) = \Phi\left(\delta_o^{pr} + \delta_d^{pr} + \beta^{pr} A_{o,d}^{2010} + X'_{o,d} \gamma^{pr}\right)$$

for the volume of FDI: in panel A, the total number of FDI relationships; in panel B, the number of firms in d which are a subsidiary of a firm in o, a measure of inward FDI; and in panel C, the total local employment in county d at subsidiaries of firms in o.

Across all specifications, we find a positive impact of ancestry on the volume of FDI. The effect of ancestry on the intensive margin of FDI, the coefficient  $\kappa$  in (7), is statistically and economically significant across most specifications: doubling the number of residents in county d who report ancestry from country o (from the mean, 316, to 632) increases the number of FDI relationships by 6.5% and local employment at subsidiaries of foreign firms by 7.3%.<sup>36</sup> More descendants of foreign migrants increases the likelihood that local firms engage in FDI, the number of firms that do so, and the local employment by foreign-owned firms.

**3.5.7. Trade versus FDI.** In Supplementary Appendix Table 20, we test whether ancestry has a similar impact on the intensive margin of FDI (panel A) as on trade (exports in panel B, and imports in panel C). We use readily available data on trade flows between U.S. states and foreign countries sourced from the U.S. Census Bureau. We again instrument for the composition of ancestry as in (4), except that all variables are defined at the state, not county, level. We correct for the selection bias due to zero trade using a Heckman (1979) correction as above.

The impact of ancestry on FDI at the state-level is positive, significant, and larger than on trade, once we instrument and include both origin and destination effects (column 3).<sup>37</sup> The effect of ancestry on trade becomes indistinguishable from zero in our preferred specification in column 3. Although we interpret this non-result with due caution due to the limited data available, it contrasts with earlier findings in the literature, started by the seminal contributions of Gould (1994) and Rauch and Trindade (2002) (using OLS), and the recent IV results of Cohen *et al.* (2015) for trade with Japan, and Parsons and Vezina (2018) for trade with Vietnam, that all find the presence of migrants facilitates exports. Our preferred specification shows no such positive and statistically significant effect. A closer look at the data suggests two important features are essential in reaching this negative conclusion: when either a formal identification is missing (OLS in column 1), or no control for destination—U.S. state—fixed effect is included (column 2), we erroneously find a positive and significant estimated impact of ancestry on trade. But when both are present (column 3 panels B and C), we find none.

**3.5.8.** Ancestry and immigration. According to our reduced-form model (2), migrations are driven by economic attractiveness (the interaction of our pull and push factors) and the stock of ancestry (the recursive factor). To provide direct evidence for these two forces, we estimate

$$I_{o,d}^{t} = \delta_{o} + \delta_{d} + \theta \ I_{o,-r(d)}^{t} \frac{I_{-c(o),d}^{t}}{I_{-c(o)}^{t}} + \lambda A_{o,d}^{t-1} + X_{o,d}^{\prime} \gamma + \vartheta_{o,d}$$
(8)

for t = 2000, 1990 (the census years for which we have information on lagged ancestry), where we again instrument for  $A_{o,d}^{t-1}$  using (4). Column 1 of Table 6, Panel A estimates (8) with immigration

with  $A_{o,d}^{2010}$  instrumented as in equation (4). We extract the predicted latent variable that determines non-zero FDI,  $\hat{z}_{o,d} = \Phi^{-1}(\hat{\rho}_{o,d})$ , and include the inverse Mills ratio  $\hat{\mu}_{o,d} = \varphi(\hat{z}_{o,d})/\Phi(\hat{z}_{o,d})$  within our set  $X_{o,d}$  of controls.

<sup>36.</sup> Using  $\hat{k} = 0.326$  in panel C, column 3 of Supplementary Appendix Table 19 in equation (7), we have:  $\frac{\text{Employment}_{o,d}[\text{Ancestry}_{o,d}=2\times316]}{\text{Employment}_{o,d}[\text{Ancestry}_{o,d}=2\times316]} - 1 = \exp\left(0.326\left(\ln\left(1+\frac{2\times316}{1000}\right) - \ln\left(1+\frac{316}{1000}\right)\right)\right) - 1 \approx 0.073.$ 

<sup>37.</sup> In unreported robustness checks, we find similar results for other years, or when restricting our analysis only to trade in manufacturing goods, where determining the final destination (origin) of an import (export) is less subject to measurement error, as well as for separate regressions for final goods and for intermediate inputs.

	(1)	(2)	(3)
Panel A	Immigration 1990–2000	Log immigration 1990–2000	Log immigration 1980–1990
Log ancestry 1990	9.662**	0.556***	
Log ancestry 1980	(4.455)	(0.075)	0.447*** (0.076)
$I_{\alpha,-r(d)}^{2000} \frac{I_{-c(o),d}^{2000}}{I_{2000}}$	1.082***	0.033**	
1-c(o)	(0.358)	(0.015)	
$I_{o,-r(d)}^{1990} \frac{I_{-c(o),d}^{1990}}{I_{1}^{1990}}$			0.061***
-c(o)	612,495	612,495	(0.015) 612,495
Panel B	FDI 20	14 (Dummy)	Log total no. of FDI relationships 2014
Log ancestry 2010	0.086***	0.082***	0.142***
FDI 2007 (Dummy)	(0.018) $0.542^{***}$ (0.031)	(0.016) 0.544*** (0.032)	(0.049)
Log Total no. of FDI Relationships 2007	(0.001)	(0.052)	0.685***
Ν	612,495	612,495	10,851

TABLE 6	
The effect of ancestry on immigration flows and FDI flow	ws

*Notes:* Panel A of the table presents the coefficient estimates from IV regressions of equation (8) at the country-county level. The dependent variable is the immigration flow from 1990 to 2000 in columns 1–2 and the immigration flow from 1980 to 1990 in column 3. In all columns, we instrument for *Log Ancestry* with the double-interactions of pull and push factors from prior censuses,  $\{I_{o,-r(d)}^{t}(I_{-c(o),d}^{t}/I_{-c(o)}^{t})\}_{t=1880,...,1980}$ . All specifications control for log distance, latitude difference, origin×destination-census-region, and destination×continent-of-origin fixed effects. Standard errors are given in parentheses and are clustered at the origin country level. Panel B: Columns 1 and 2 present coefficient estimates from IV regressions of equation (1) at the country-county level analogous to Table 3 column 3. The dependent variable is a dummy indicating an FDI relationship between origin o and destination d in 2014. Column 1 of panel B includes all double-interactions as instruments,  $\{I_{o,-r(d)}^{t}(I_{-c(o),d}^{t}/I_{-c(o)}^{t})\}_{t=1880,...,2000}$ , while columns 2 and 3 include only double-interactions of the push and pull factors from the 1990 and 2000 censuses,  $\{I_{o,-r(d)}^{t}(I_{-c(o),d}^{t}/I_{-c(o)}^{t})\}_{t=1990,2000}$ . Panel B column 3 presents the IV/GMM estimate of equation (7). This specification controls for log distance, latitude difference, origin, and destination × continent, and origin × census region fixed effects. Standard errors are given in parentheses and are clustered at the origin country level. \*, \*\*, and \*\*\* denote statistical significance at the 10%, 5%, and 1% levels, respectively.

 $I_{o,d}^t$  in levels, and (reassuringly) gives a coefficient on the interaction of the push and pull factors close to 1. Columns 2 and 3 estimate (8) in logs for two time periods, 1990 and 2000. Across all specifications, both the coefficient on the push  $\times$  pull interaction and on lagged ancestry are positive and significant predictors of current migrations.

In Panel B of Table 6, we show recent changes in FDI are predicted by recent changes in ancestry. In column 1, we add a dummy for FDI in 2007 (the earliest year for which we have data) to our standard specification from column 3 of Table 3, so that the coefficient of interest now reflects the effect of ancestry on *changes* in the extensive margin of FDI between 2007 and 2014.<sup>38</sup> As expected, we find a smaller, but nevertheless positive coefficient (0.086, SE = 0.018). Column 2 shows almost identical results when using only migrations between 1980 and 2000 for identification. Column 3, shows the same effect at the intensive margin (0.142, SE = 0.049).

38. The magnitude of this effect should be interpreted with caution as coverage of the ORBIS data has expanded between 2007 and 2014 so that the size of the effect will reflect both changes in FDI and changes in coverage.

#### 3.6. Quantifying the effect of ancestry on FDI

We illustrate the quantitative implications of our findings using two thought experiments. First, we estimate how FDI linkages with China might have evolved if Chinese migrants had not been effectively barred from entering the U.S. between 1882 and 1965. Second, we report how FDI in Los Angeles might have evolved if Los Angeles had had an influx of migrants in the 1800s similar to that resulting from the San Francisco Gold Rush. These thought experiments are not meant as formal counterfactuals, but merely as illustrations of the magnitude of the long-term effect of immigration policies on FDI implied by our estimates.

**3.6.1.** The effect of Chinese exclusion. The U.S. government passed the Chinese Exclusion Act into law in 1882 in response to increased immigration from China, essentially closing the U.S. to legal immigration of labourers from China. In 1943, it was replaced by the Magnuson Act, which allocated a quota of 105 immigrants per year from China, and was in effect until 1965, when the removal of the quota system allowed for large-scale Chinese immigration for the first time. We refer to the entire period from 1882 through to 1965 as the period of "Chinese Exclusion". How different would the ancestry composition and FDI of U.S. counties be today had it not been for Chinese Exclusion?

To answer this question we require an estimate for the impact of Chinese exclusion on the number of immigrants from China. We use our own data to derive a rough estimate. We aggregate our immigration data at the time × census-region × origin level and estimate  $I_{o,r}^t = \delta_{t,r} + \delta_o - \hat{\xi} \cdot D_{\text{China}}^t + v_{t,o,r}$ , where  $D_{\text{China}}^t$  is a dummy equal to 1 if o=China and  $t \in [1882, 1965]$ , and  $\delta_{t,r}$  and  $\delta_o$  are time × census region and origin fixed effects, respectively. The coefficient  $\hat{\xi}$  can then be interpreted as the estimated average negative impact of the Chinese Exclusion Act on Chinese migrations. We then calculate a hypothetical time path of immigration in the absence of Chinese exclusion as  $\tilde{I}_{o,r}^t = I_{o,r}^t + \hat{\xi} \cdot D_{\text{China}}^t$ . It suggests that the U.S. would have received 1.8 million additional Chinese immigrants during the period of exclusion.

Given this hypothetical time path of immigration we can then use our estimates from Table 2 to predict the change in ancestry as  $dA_{o,d} \equiv \sum_t \hat{\alpha}_t \cdot (\tilde{I}_{o,-r(d)}^t - I_{o,-r(d)}^t) \frac{I_{-c(o),d}^t}{I_{-c(o)}^t}$ , where  $\hat{\alpha}_t$  are the estimated first-stage coefficients. The hypothetical change of FDI relations with China at the county level is  $d\Pr[\text{FDI}_{o,d} > 0] \equiv \hat{\beta} \cdot dA_{o,d}$ , where  $\hat{\beta}$  is the estimated second-stage coefficient in a specification as in column 3 in Table 3, excluding the principal components to be consistent with the above described methodology to predict hypothetical levels of ancestry.

These calculations suggest that the increase in Chinese migration would have been highly unequally distributed, translating into heterogenous changes in the incidence of FDI relationships with China. The map in Figure 6 depicts the expected change in the probability of positive FDI with China,  $dPr[FDI_{China,d} > 0]$ . The absence of Chinese exclusion would have resulted in substantially stronger FDI ties with the Northeast, the Midwest and the Southwest. The bar graph depicts the fraction of counties within a state which have positive FDI with China in 2014, and the predicted change in this measure of the extensive margin of FDI linkages, i.e. the unweighted average of  $dPr[FDI_{China,d} > 0]$  across counties within the state. To save space, the graph shows only the ten states with the highest predicted change. For example, we predict that in the absence of Chinese exclusion, the proportion of counties with an FDI link to China would have doubled in New York, and increased by 60% in Massachusetts and Illinois.<sup>39</sup>

<sup>39.</sup> Although we believe, in light of our results, that additional Chinese immigration likely also raises FDI between China and the U.S. as a whole, we cannot exclude the possibility that increases in FDI with China in one state are partially or fully offset by decreases in others.



#### FIGURE 6

Thought Experiment: Removing the Chinese Exclusion Act. *Notes:* The map on the left depicts for each US county the predicted increase in the probability of having positive FDI relations with China in a counterfactual world where the "Chinese Exclusion" Act of 1882 had never been passed. Darker colours indicate larger increases. The bar graph on the right shows the fraction of counties within each state with FDI relations with China (light colour) and the predicted increment in the fraction of counties with FDI relations with China (dark colour), which we calculate as the unweighted average of  $d\Pr[FDI_{China,d} > 0]$  across counties for the ten US states with the largest change. We also provide the size of this increase relative to the actual fraction in percentage terms. The details of this calculation are Section 3.6.1

3.6.2. A gold rush in Los Angeles. To similarly gauge the magnitude of the estimated intensive margin effects, we derive predictions for the intensity of FDI relationships between Los Angeles county and the world under the hypothetical scenario that Los Angeles had experienced a Gold Rush similar to that in San Francisco, a fivefold increase in the pre-1880 number of immigrants to Los Angeles. Supplementary Appendix Table 21 presents the results of this thought experiment for the 10 foreign countries with the largest predicted change in their ancestry group in Los Angeles in 2010. Columns 1 and 2 show the actual number of individuals of each ancestry in Los Angeles County in 2010 and the total number of FDI links recorded in our data between Los Angeles County and the respective origin countries. Columns 3 and 4 present the predictions of our thought experiment based on the IV specification corresponding to column 2 of Supplementary Appendix Table 19, again without the principle components as instruments. A Gold Rush in Los Angeles would have resulted in sizeable effects on the intensity of FDI with those countries that were the source of immigration pre-1880: the intensity of FDI between Los Angeles county and Germany and Ireland would have increased by around 60%. Column 4 presents the predicted absolute change in the size of the ancestry groups, based on a reduced form regression analogous to column 9 of Table 2 with Ancestry 2010 (in levels) as outcome variable, again excluding the principle components. It suggests that the population of Irish and German descent living in Los Angeles county today would each be counting about 60,000 more individuals.

## 4. UNDERSTANDING THE EFFECT OF ANCESTRY

So far, we have documented a quantitatively large causal effect of common ancestry on FDI. We now turn to the mechanism linking ancestry to FDI. Existing research suggests that migrations and common ancestry may affect FDI either by making the destination more "similar" to the

origin in terms of preferences and skill endowments or by generating social capital that creates an absolute advantage for firms to operate in both the origin and destination country.

In the first category, Atkin (2013) and Bronnenberg *et al.* (2012) suggest that descendants of migrants may share the same tastes for foods and other products as consumers in their origin country. To the extent that these tastes persist over generations, firms that cater to those tastes may serve both markets. Similarly, we might suspect that migrants may bring with them a specific skill-mix or other factors abundant in their origin country, so that firms can more easily outsource production, using the same skill-mix at home and abroad.

In the second category, common ancestry may create an absolute advantage in conducting FDI for local firms, because social ties between populations in the origin and the destination provide social collateral that helps to enforce contracts when the legal system of *o* or *d* is imperfect (Greif, 1993; Besley and Coate, 1995). Alternatively, migrants and descendants of migrants from a given origin may have privileged access to information (Varian, 1990; Stiglitz, 1990): a more intimate knowledge of the business environment in their origin country and social ties or language skills that provide access to information about business opportunities and practices at a lower cost.

The following section presents evidence testing auxiliary predictions of these distinct channels. The collection of these results suggests that ancestry affects FDI primarily because it creates an absolute advantage for local firms by attenuating information frictions.

#### 4.1. Channel linking ancestry to FDI

Panels A–C of Table 7 present indirect tests for the "similarities" hypothesis. Panel A shows the IV coefficient of ancestry on FDI separately for firms producing final goods and for firms producing intermediate inputs.<sup>40</sup> If common tastes were the explanation behind the positive impact of ancestry on FDI, we would expect its impact to be stronger for final goods, for which consumers' tastes matter directly, than for intermediate inputs, for which tastes matter little. We find, on the contrary, that there is no significant difference between final goods and intermediate input producers; if anything, the point estimate is slightly larger for intermediate input producers than for final goods producers. Panel B shows similar results for inward FDI only, where the local tastes of descendants from country *o* may plausibly matter more.

Panel C of Table 7 shows the IV coefficients of a regression of ancestry on measures of sectoral similarity between the origin and the destination. For each origin-destination pair, we compute the rank and cosine correlation of the shares of employees in 127 manufacturing sectors.<sup>41</sup> Both correlations increase with the similarity of the allocation of employees across sectors between the origin and destination. If skill similarities were the explanation behind the positive impact of ancestry on FDI, we would expect common ancestry to cause an increase in these measures. We find, on the contrary, that ancestry has no discernible impact on sectoral similarity. This non-result—migrations do not cause a convergence in the sectoral composition of employment—is robust to using alternative measures of sectoral similarity, as well as alternative data sources.<sup>42</sup>

Panel D of Table 7 examines the contract enforcement channel. If contract enforcement were the explanation behind the positive impact of ancestry on FDI, we would expect the impact to

42. Results are unchanged whether we use rank or cosine correlation, or when we repeat the same exercise using the OECD Stan country level industry data.

<sup>40.</sup> We use the upstreamness index from Antràs *et al.* (2012) to define sectors producing final versus intermediate goods: a sector is labelled as final goods (intermediate input) if its upstreamness index is below (above) 2.

<sup>41.</sup> We use county and country level industry data from the Bureau of Labor Statistics (BLS) and the UN Industrial Development Organization (UNIDO), respectively. Correlations are calculated for 2006, the year with the largest availability of data (twenty-eight countries). Using this smaller sample of countries, the coefficient on ancestry in our standard specification linking ancestry to FDI is 0.348 (SE = 0.046).

#### **REVIEW OF ECONOMIC STUDIES**

	(1)	(2)
Panel A: final versus intermediate goods	FDI 2	014 (Dummy)
Log ancestry 2010	0.156***	0.169***
C .	(0.026)	(0.024)
Ν	612,495	612,495
Sample	Final goods	Intermediate goods
Panel B: final versus intermediate goods	Inward F	DI 2014 (Dummy)
Log ancestry 2010	0.108***	0.117***
	(0.033)	(0.032)
Ν	612,495	612,495
Sample	Final goods	Intermediate goods
Panel C: sector similarity	Rank correlation	Cosine correlation
Log ancestry 2010	0.011	0.010
	(0.015)	(0.013)
Ν	21,518	21,518
Panel D: judicial quality	FDI 2014 (Dummy)	Log no. of FDI relationships
$Log$ ancestry $\times$ Judicial quality	0.180*	1.414***
• •	(0.094)	(0.243)
Ν	452,304	10,089

		TABLE 7		
The	"similarities"	hypothesis and	l contract	enforcement

*Notes:* The table presents coefficient estimates from IV regressions at the country-county level. In Panel A, the outcome variable is the FDI dummy; we restrict our sample to firms producing final goods or intermediate inputs, respectively. Final goods and intermediate inputs are defined as four-digit NAICS sectors with upstreamness index below and above 2, respectively, where we use the upstreamness index from Antràs *et al.* (2012). The number of country–county pairs that have an (non-zero) FDI link in the corresponding sector is 4,201 and 5,842 in columns 1 and 2, respectively. In Panel B, we replicate the same regressions, except that the outcome variable indicates only the existence of any inward FDI. In Panel C, the outcome variable is the rank and cosine correlation of the share of employees in 127 manufacturing sectors within a given origin-destination pair, respectively. The relatively low number of observations is due to data availability in the industry share of employment: When calculating the correlation between industries' share of employment in county *d* and country *o*, the correlation coefficient is missing for those country–county pairs that have at least one missing share of employment. In Panel D, the outcome is the extensive (FDI dummy) and intensive (log no. of FDI relationships) margin, and the measure of judicial quality is from Nunn (2007). Throughout we use  $\{I_{o,-r(d)}^{t}(I_{-c(o),d}^{t}/I_{-c(o)}^{t})\}_{t=1880,...,2000}$  and principal components as instrumental variables. All specifications control for log distance, latitude difference, and origin and destination fixed effects. In Panels A–C we additionally control for origin×destination-census-region, and destination×continent-of-origin fixed effects. Standard errors are given in parentheses and are clustered by origin country.

be stronger for countries where the quality of the local judiciary is weaker, as ethnic ties would substitute for weak institutions. We find the opposite result in columns 1 (extensive margin) and 2 (intensive margin), where we add the interaction of ancestry with a measure of the origin country's judicial quality taken from Nunn (2007) to our simple IV specification of column 2 in Table 3. The coefficients show that the effect of ancestry on FDI is significantly larger for countries with good institutions than for countries with bad institutions, suggesting that common ancestry and good institutions are complements rather than substitutes.

We conclude from Table 7 that the data show no evidence for the "similarities" hypothesis and that our results do not seem driven by ethnic ties substituting for poor contract enforcement.

#### 4.2. Information demand

To directly test the remaining hypothesis that common ancestry affects FDI by reducing information frictions, we require data on the flow of information from foreign countries to U.S. destinations. Because such data are not readily available we construct a simple index of differential

information demand using data from internet searches.<sup>43</sup> The Google Trends portal provides data on the relative popularity of different search terms across 210 U.S. metropolitan areas ("media markets" according to the Nielsen DMA definition).<sup>44</sup> For each search term *i* and U.S. metro area *d*, the portal returns an index number that is equal to the normalized share of searches conducted in *d* that contain the search term *i*:

$$G(i,d) = \left[100 \frac{\text{share}(i,d)}{\max_{\delta}\{\text{share}(i,\delta)\}} \mathbf{1}[\#(i,d) > T]\right],$$

where share(*i*,*d*) is the share of searches in *d* that contains *i* and  $\mathbf{1}[\#(i,d) > T]$  is an indicator function that is one if the absolute number of searches containing *i* in *d* is greater than some threshold number (Stephens-Davidowitz and Varian, 2015; Liang, 2017).<sup>45</sup> Thus, *G*(*i*,*d*) is equal to 100 in the metro area in which the largest share of searches contain *i* and a positive number smaller than 100 in all other metro areas that have a sufficient number of searches containing *i*.

To measure the relative demand in a given metro area for information about a given origin country, we compile a list of the five most prominent actors, athletes, musicians, and politicians for each origin country. We automate this process by searching for "notable [country] [category]" and then extracting the five top suggested names from the Google Answer Box, a feature of Google search that automatically suggests the most often clicked names associated with this kind of query.<sup>46</sup> We then calculate our Information Demand Index as

$$\text{IDI}(o,d) = \frac{1}{20} \sum_{p} \sum_{i \in q(o,p)} G(i,d),$$

where q(o,p) is the set of top five names for country o in category  $p \in \{ \text{ actors, athletes, } ... \}$ . We implement this procedure for the 100 largest foreign countries by 2015 population. To facilitate the interpretation of results, we standardize this measure to a unit standard deviation.

Panel A of Table 8 shows the results of regressions of our differential information demand index on ancestry (instrumented as in (4)), and our standard set of controls at the metropolitan area — country level. Column 1 documents a large causal effect of ancestry on information demand (0.871, SE = 0.257), where doubling the number of descendants of migrants from a given origin relative to the mean is associated with a 0.19 standard deviation increase in our index of demand for information about prominent actors, athletes, musicians, and politicians in that origin.<sup>47,48</sup> Columns 2 and 3 show that this effect remains positive and statistically highly significant even when we control for the foreign-born population (first-generation immigrants), and when we

43. We thank Jack Liang for writing his Bachelor Thesis at the University of Chicago on this topic.

47. Following the same calculation as above we have  $0.871\left(\ln\left(1+\frac{632}{1000}\right)-\ln\left(1+\frac{316}{1000}\right)\right)=0.19$ .

48. Complimentary evidence to ours is provided in Bailey *et al.* (2016), who find that recent immigrations from origin country o to U.S. county d, as well as the composition of ancestry across U.S. counties d, are close correlates of a measure of social ties derived from Facebook friendship links.

<sup>44.</sup> For other recent studies using this data source see Da et al. (2011), Stephens-Davidowitz (2014), Kearney and Levine (2015), and Baker and Fradkin (2016).

<sup>45.</sup> As a result of this cutoff, our index tends to assign a value of zero to small origin-destination pairs (38% of our sample). For this reason, we focus our attention on the 100 largest origin countries by 2015 population and do not attempt to construct it for all present-day origin countries.

<sup>46.</sup> See Supplementary Appendix Table 22 for the full list of search terms used for Germany and Italy as examples. Liang (2017) shows evidence that search terms with multiple meanings (*e.g.* two prominent politicians from two different countries share the same name) do not impact our results, and gives a detailed account demonstrating that the Google Answer Box generally delivers relevant search terms for each country. See Supplementary Appendix A.4 for details.

	The effe	ect of ancestry on c	TABLE 8 lifferential inforr	mation demand and la	nguage		
Panel A: google trends	Information de (standar	emand index dized)		Actors (standardized)	Athletes (standardized)	Musicians (standardized)	Politicians (standardized)
	(1)	(2)	(3)	(4)	(5)	(9)	(7)
Log ancestry 2010	0.871***	1.717**		0.659**	0.948***	0.531***	0.688***
Log foreign-born 2010	(1 67.0)	(0.00) -1.108 (0.000)		(107.0)	(600.0)	(0/0/0)	(CQ1.0)
Log ancestry 1980		(600.0)	0.801***				
Ν	19,110	19,110	(0.100) 19,110	19,110	19,110	19,110	19,110
Destination FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Origin FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Principal components	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Panel B: language	No. of residents in <i>d</i> speaking language of <i>o</i> at home			No. of US the lang	-born in $d$ that speak uage of $o$ at home		
	(1)	(2)	(3)	(4)	(5)	(9)	(2)
Ancestry 2010	2.226***	0.942***		$1.170^{***}$	1.257***	0.085***	0.015***
Ancestry 1980	(0./1/)	(105.0)	$1.241^{**}$	(0.042)	(0.230)	(0000)	(500.0)
Ν	454,812	454,813	(0.504) 454,813	65,877	78,376	3,137	3,137
Non-English language	Any	Any	Any	Spanish	Arabic	Chinese	Hindi
Destination FE	Yes	Yes	Yes	No	No	No	No
Origin FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
<i>Notes</i> : This table presents on a count of Google sear four categories is the aver of Designated Market Are outcome variables in Panel home, excluding English; i US-born residents in <i>d</i> that countries; Chinese and Hi are instrumented as in coll are instrumented as in coll	coefficient estimates from IV regress thes in the category specified above gae of Google Trends value $G(i, d)$ , a (DMA) $d$ . For the search terms $i$ w A are standardized by their standard n columns 2 and 3, it is the number o speak the language indicated in the 1 di each are the official language in dime 2 of Table 3. All specifications	sions at the countr the panel. The Infu the panel. The Infu we use the first five deviation. In Pane of US-born resident respective column. only one country.	y-DMA (Panel , primation Deman the (normalized) terms from Goo el B, the depende s in $d$ that speak Spanish is the o Ancestry 2010, , the astree and latitu	A) and country-country di Index in columns 1 di fraction of queries th ogle's Answer Box wl part variable in column the language of o at h dificial language in two Ancestry 1980, Log A Ancestry 1980, Log A ande difference. Standa	y (Panel B) level. In P -3 is a simple average at include search term hen we search for "noi 1 is the number of res ome, excluding Englis anty-one countries; Arr uncestry 2010, Log Fo	anel A, all dependent s of the other four cata i <i>i</i> relative to the total table foreign country idents in <i>d</i> that speak h; and in columns $3-6$ h; and in columns $3-6$ reign-born 2010, and parenteses and clust	variables are based variables are based mumber of queries d] [category]". All the language of $o$ at , it is the number of uage in twenty-five Log Ancestry 1980 ered at the country
TOVEL (ALL VI LATER LA MINT A	1 - 1 = 1 = 1 = 1	T (VUILINE J- V VI	1 duvi 1	, allu uviiviv oran	ouval organizative as a	IIC 10/0, J/0, mIL 1/0	ICVUID, LUDPULLI VIJ.

32

# **REVIEW OF ECONOMIC STUDIES**

condition only on ancestry in 1980 (rather than 2010). Taken together, these results suggest that the differential interest in information about the origin country persists among the U.S.-born descendants of first-generation migrants.

The remaining columns show that this persistent interest in the origin country is not limited to politics, but is similar across our four sub-indices for demand for information about actors, athletes, musicians, and politicians.

The longevity of the effect of ancestry on differential information demand suggests that immigrants pass traits to their descendants that facilitate or encourage the exchange of information with their origin countries, such as social ties to family or friends, or knowledge of the origin country's language and culture. Although data on such traits is generally hard to come by at the required level of disaggregation, Panel B shows one additional piece of evidence from the U.S. census: the use of foreign languages. The table shows systematic evidence that a larger community in county d with ancestry from country o has a positive and significant impact on the number of residents in d who speak o's language at home (column 1). This effect persists if we remove from d's population all foreign-borns, since they "mechanically" speak the foreign language from their home country (column 2). Columns 4–6 present the results from separate regressions for large non-English languages: Spanish, Arabic, Chinese, and Hindi.<sup>49</sup> The effect of ancestry on foreign language spoken is positive and statistically significant for all four.

# 4.3. Ancestry, information flow, and FDI

We next ask whether these differences in information flows can account for the the link between ancestry and FDI. We get a superficial answer to this question in Supplementary Appendix Table 24 by running our simple specification relating ancestry to FDI (column 2 in Table 3), while controlling for our information demand index. We find that the coefficient on ancestry drops to close to zero and becomes statistically insignificant (-0.025, SE = 0.028), while the coefficient on information demand remains positive and highly statistically significant (0.078, SE = 0.013). In contrast, controlling for sectoral similarity and the various other channels probed above has no effect on the causally identified coefficient on ancestry. Our results thus suggest the effect of ancestry on FDI transits through the information channel. We study next the mechanisms through which information transmission may generate an absolute advantage in conducting FDI.

**4.3.1.** Network effects of common ancestry. Theoretical models emphasize the role of networks in facilitating the percolation of information across international borders (Arkolakis, 2010; Chaney, 2014). This class of models tends to predict effects that are concave (as all the relevant information is gradually exhausted), weaker if many people from the same or neighbouring origins live in the surrounding area (as relevant information is more likely to have already percolated), and stronger for destinations that are more ethnically diverse (hubs open up more paths for information to percolate through). We test each of these reduced-form predictions.

We have already shown in Section 3.5 that the relationship between ancestry and FDI is concave (see Supplementary Appendix Figure 6). Information percolation on a network also suggests negative spillovers from neighboring regions: if many people with ancestry from o live in locations surrounding d, or if many people in o have an ancestry from countries adjacent to o, it is more likely that relevant information about investment opportunities has already reached the

<sup>49.</sup> Supplementary Appendix Table 23 presents the regression coefficients of ancestry on foreign languages for the fifty largest linguistic groups.

### **REVIEW OF ECONOMIC STUDIES**

firm, so that the marginal impact of ancestry on FDI is mitigated. In Table 9, panel A, column 1, we use our simple specification from column 2 in Table 3, but add the total number of descendants of ancestry o at the state level. We are able to identify the effect of this spillover at the state level by aggregating our instruments from equation (4) to the state level and including them as a separate set of instruments, such that both endogenous variables are identified. The coefficient on our measure of ancestry at the state level is -0.015 (SE = 0.004), suggesting a negative and significant spillover. In column 2, we include a measure of the number of descendants from the closest neighboring country, and we again find a negative and highly significant spillover effect.<sup>50</sup> Our findings are thus consistent with the presence of negative spillovers within co-ethnic networks.

In columns 3 and 4, we repeat the same estimation for the intensive margin of FDI, but in this much smaller sample we lack the statistical power to identify a consistent pattern.

An additional prediction of network-based models is the existence of hubs, dense locations through which distant locations can connect. If such hub-effects were at work, we would expect the effect of ancestry on FDI to be larger in more ethnically diverse destinations. We test this prediction in panel B of Table 9. We interact ancestry with a measure of ethnic diversity (measured as 1 minus the Herfindhal index of ancestry shares). The coefficient on this interaction is positive and significant, both at the intensive and the extensive margin (columns 2 and 4). In contrast, we find no effect on the interaction between ancestry and the share of the population that are of foreign descent (columns 1 and 3). This suggests the effect in columns 2 and 4 is driven by ethnic diversity, not by the size of the population share with foreign descent.

We conclude that the patterns by which ancestry affects FDI are consistent with the auxiliary predictions of models of network effects, where information (or other effects of social capital) are transmitted internationally through networks created by common ancestry.

**4.3.2.** Cost of information transmission. If information frictions indeed were the explanation behind the positive impact of ancestry on FDI, we would also expect this effect to be stronger in relationships that suffer from higher costs of information transmission. For example, information costs may be larger for distant countries or countries that are ethnically diverse.

We confirm this prediction in the Panel C of Table 9. In all specifications, the coefficient on the interaction between ancestry and geographic distance is positive and significant, both for the extensive (columns 1 and 2) and intensive margins of FDI (columns 3 and 4).<sup>51</sup> Columns 2 and 4 also show some evidence that the effect of ancestry on FDI is larger for more ethnically diverse origins (a higher level of ethno-linguistic fractionalization as defined by Alesina *et al.* (2003)).<sup>52</sup>

# 4.4. Generational effects

Having already shown in Section 3 that historical migrations prior to World War II had causal effects on FDI that persist to the present day, and that historical migrations predict future migrations through a recursive factor, we now ask whether the effect of ancestry on FDI requires a sustained inflow of migrants from the same origin—or if it persists even after migration from

<sup>50.</sup> We determine the nearest adjacent country by creating country pairs, using a standard optimal non-bipartisan matching algorithm, such that the average distance between centroids of country pairs is minimized.

<sup>51.</sup> Once we account for this interaction, the interaction terms on genetic, linguistic, and religious distance, as defined by Spolaore and Wacziarg (2016), are statistically insignificant, suggesting geographic distance effectively summarizes alternative notions of distance in cultural space.

<sup>52.</sup> Results are virtually identical when we consider outward FDI by itself.

Network effects					
	FDI 2014 (Dummy)		Log total no. of FDI relationships		
	(1)	(2)	(3)	(4)	
Panel A: spillovers					
Log ancestry 2010	0.237***	0.166***	1.027***	0.179***	
	(0.020)	(0.026)	(0.168)	(0.052)	
Log ancestry 2010, state level	-0.015***	· /	-0.315***		
	(0.004)		(0.052)		
Log ancestry 2010 of nearest origin country	(0.001)	-0.047**	(0.052)	0.037	
		(0.019)		(0.176)	
N	612 405	(0.01))	10.851	10.851	
1v	012,495	012,495	10,651	10,651	
Panel B: diversity					
Log ancestry 2010	0.197***	0.123***	0.661***	0.641***	
	(0.061)	(0.030)	(0.245)	(0.089)	
Log ancestry $\times$ foreign share	1.388		3.548		
	(3.103)		(7.803)		
Log ancestry $\times$ ethnic diversity	(00000)	1.270***	(,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	3 692***	
		(0.204)		(1.009)	
Ν	611,910	612,495	10,851	10,851	
Panel C: fractionalization					
Log ancestry 2010	0.269***	0.334***	0.914***	1.247***	
	(0.037)	(0.081)	(0.149)	(0.109)	
Log ancestry $\times$ geographic distance	0.101***	0.170**	0 414***	0.864***	
	(0.036)	(0.076)	(0.128)	(0.156)	
Log ancestry × judicial quality	(0.050)	0.373**	(0.120)	2 375***	
Log uncestry × judicial quality		(0.197)		(0.404)	
Log ancestry $\times$ fractionalization		(0.187)		2.007***	
		0.470		3.08/	
N7	116.000	(0.324)	10.000	(0.831)	
N	446,022	446,022	10,089	10,089	

TABL	E 9
M 1	. CC

*Notes:* The table presents coefficient estimates from IV regressions at the country-county level. The dependent variable in columns 1 and 2 is the dummy for FDI in 2014. The dependent variable in columns 3 and 4 is the log of the number of FDI links in 2014. We use  $\{I_{o,-r(d)}^{t}/I_{-c(o)}^{t}/I_{-c(o)}^{t}\}_{t=1880,...,2000}$  and principal components as instruments. All specifications control for log distance, latitude difference, origin, and destination fixed effects, as in column 2 of Table 3. In columns 2 and 4 of Panel A, destination × continent fixed effects are used to control for the increased substitutability of immigrants from a common continent. Foreign Share, Ethnic Diversity, Distance, Judicial Quality, and Fractionalization are demeaned. Standard errors are given in parentheses and are clustered at the origin country level. \*, \*\*, and \*\*\* denote statistical significance at the 10%, 5%, and 1% levels, respectively. Foreign Share is the share of the destination county's population that are of any foreign ancestry in 2010. Diversity of Ancestries is measured as 1 minus the Herfindhal index of ancestry shares in the destination county. Judicial quality in the origin is from Nunn (2007); genetic distance is from Spolaore and Wacziarg (2016); and Ethnic Fractionalization refers to 1 minus the Herfindahl index of ethnicities in the origin country calculated using the data in Alesina *et al.* (2003).

that origin ceases. Supplementary Appendix Table 25 compares the (causally identified) effect of ancestry to that of foreign born, that is, first-generation immigrants. Column 1 replicates our standard specification for comparison. Column 2 replaces our measure of ancestry in equation (1) with a measure of foreign born from a given origin alive in 2010, instrumenting as in equation (4). The effect remains positive and significant (unsurprisingly, as the correlation between the two variables is 0.59). When we simultaneously include both endogenous variables in the specification, the coefficient on ancestry remains positive and statistically significant at the 1% level, whereas the coefficient on foreign born in 2010 is close to zero and insignificant in the OLS specification in column 3. In the IV specification in column 4 the coefficient turns slightly negative and marginally statistically significant. This suggests the presence of the descendants of immigrants continues to predict FDI even after migrations from the origin cease. Using the number of foreign born in 1970 as a proxy for second-generation immigrants, column 6 compares the marginal effect of second-generation immigrants to that of the average resident with foreign ancestry. The coefficient on second-generation immigrants remains positive (albeit not significant) when we control for descendants of migrants.<sup>53</sup> Although these specifications, disentangling the marginal effects of several endogenous variables, should be interpreted with caution, they suggest the effect of ancestry on FDI develops over long periods of time, and possibly peaks with the second, but not the first generation of immigrants.

This finding is consistent with a set of microeconomic studies that show that only those individuals that advance to managerial positions successfully establish business linkages to their origin countries (Aleksynska and Peri, 2014); and that it tends to be the second and third generations of immigrants, that achieve such advancement (Borjas, 2006; Algan *et al.*, 2010). Simply put, the second generation of immigrants is more likely than the first to be able to act on any privileged access to information about the origin country.

To conclude, we find a collage of evidence that migration, and the distribution of ancestry that results from it, has a positive impact on FDI primarily because it reduces information frictions associated with foreign direct investment. We also find evidence consistent with network effects and the inter-generational transmission of traits that facilitate the flow of information between the origin country and the U.S. destination on a long-term basis.

# 5. CONCLUSION

The economic effects of migration loom large in public debates about illegal immigration to the U.S. and the ongoing flow of migrants to Europe from places such as Syria, Afghanistan, Africa, and the Balkans. Much of the academic debate on the subject has focused on relatively short-term consequences on local labour markets and consumer prices (Card, 1990; Cortes, 2008). We contribute to this debate by showing causally identified evidence that migrations, and the ethnic diversity they create, also have a long-term effects: They enhance the propensity of firms based in areas receiving migrants to interact economically with the migrants' origin countries. This ethnic determinant of foreign direct investment is large, persists for generations, and seems to operate primarily through a reduction in information frictions.

With these findings, we shed new light on the economic effects of migrations, and suggest several promising avenues for future research. First, we document that ethnic diversity increases the likelihood and intensity of FDI. Because the effect of foreign ancestry on FDI is strongly concave, receiving migrants from many small ethnic groups generates more FDI than receiving the same number of migrants from a single ethnic group. In this sense, ethnic diversity creates an absolute advantage in conducting FDI. A similar argument could potentially be made for the impact of ethnic diversity on the adoption of foreign technologies or on economic growth.

Second, the effects of historical migrations on economic development are large and longlasting. Regions within the U.S. that received more, often poor, migrants from countries like Ireland or China more than a century ago, as a result, enjoy significantly stronger economic ties with these countries today. Taking in migrants today may thus deliver long-term dividends: International investments follow the paths of historical migrants as much as they follow economic fundamentals, such as productivity, wage differentials, and tax breaks.

Finally, our identification strategy has a range of applications beyond FDI. For example, our approach to identification could be used to study the broader impact of migrations on attitudes towards foreigners of various origins or the effects of migration on long-term economic growth.

<sup>53.</sup> These results hold if we drop migrations from Mexico (the largest origin country in recent decades).

If global migration pressures increase further, *e.g.* due to global climate change, having a detailed understanding of these impacts is key to devising effective migration policies.

Acknowledgments. We are grateful to Lorenzo Casaburi, Joshua Gottlieb, Richard Hornbeck, Nathan Nunn, Emir Kamenica, Jacopo Ponticelli, Nancy Qian, and David Strömberg for helpful discussions. Comments from the editor, Gita Gopinath, and four anonymous referees have helped to improve this article. We also thank seminar participants at the Barcelona GSE, Boston University, Boston College, CEPR ERWIT, University of Chicago, Columbia University, Oxford, Georgetown, Harvard, IFN (Stockholm), Imperial, the University of Maryland, MIT, NBER EEG and Culture & Institutions program meetings, Paris School of Economics, Princeton, Singapore Management University, National University of Singapore, Toulouse, UPF, and the University of Zürich for their comments. T.C. and T.A.H. are grateful for financial support from NSF grant SES-1061622. T.C. is grateful for financial support from ERC grant N°337272–FiNet. T.A.H. is grateful for financial support from the IGM and the Fama-Miller Center at the University of Chicago. Mathias Iwanowsky, Markus Schwedeler, Lisa Tarquino and Philip Xu provided excellent research assistance. All mistakes remain our own.

#### Supplementary Data

Supplementary data are available at Review of Economic Studies online.

#### REFERENCES

- AGER, P. and BRÜCKNER M. (2013), "Cultural Divsersity and Economic Growth: Evidence from the US During the Age of Mass Migration", *European Economic Review*, **64**, 76–97.
- ALEKSYNSKA, M. and PERI G. (2014), "Isolating the Network Effect of Immigrants on Trade", *The World Economy*, **37**, 434–445.
- ALESINA, A., DEVLEESCHAUWER, A., EASTERLY, W. ET AL. (2003), "Fractionalization", Journal of Economic Growth, 8, 155–194.
- ALESINA, A., HARNOSS, J. and RAPOPORT, H. (2016a), "Birthplace Diversity and Economic Prosperity" (Working Paper).
- ALEŠINA, A., MICHALOPOULOS, S. and PAPAIOANNOU, E. (2016b), "Ethnic Inequality", Journal of Political Economy, 124, 428–488.
- ALGAN, Y., DUSTMANN, C., GLITZ, A. ET AL. (2010), "The Economic Situation of First and Second-Generation Immigrants in France, Germany and the United Kingdom", *The Economic Journal*, **120**, F4–F30.
- ANTRÀS, P., CHOR, D., FALLY, T. ET AL. (2012), "Measuring the Upstreamness of Production and Trade Flows", American Economic Review Papers and Proceedings, 102, 412–416.
- ARKOLAKIS, C. (2010), "Market Penetration Costs and the New Consumers Margin in International Trade", Journal of Political Economy, 118, 1151–1199.
- ARKOLAKIS, C., COSTINOT, A. and RODRÍGUEZ-CLARE, A. (2012), "New Trade Models, Same Old Gains?" American Economic Review, 102, 94–130.
- ASHRAF, Q. and GALOR, O. (2013), "The "Out of Africa" Hypothesis Human Genetic Diversity, and Comparative Economic Development", *American Economic Review*, **103**, 1–46.
- ATKIN, D. (2013), "Trade, Tastes and Nutrition in India", American Economic Review, 103, 1629–1663.
- BAILEY, M., CAO, R., KUCHLER, T. ET AL. (2016), "Measuring Social Connectedness" (Working Paper 23608, National Bureau of Economic Research).
- BAKER, S. R. and FRADKIN, A. (2016), "The Impact of Unemployment Insurance on Job Search: Evidence from Google Search Data", *The Review of Economics and Statistics*, **99**, 756–768.
- BARTIK, T. J. (1991), Who Benefits from State and Local Economic Development Policies? (Kalamazoo MI: Upjohn Press, Institute for Employment Research).
- BESLEY, T. and COATE, S. (1995), "Group Lending, Repayment Incentives and Social Collateral", Journal of Development Economics, 46, 1–18.
- BHATTACHARYA, U. and GROZNIK, P. (2008), "Melting Pot or Salad Bowl: Some Evidence from US Investments Abroad", *Journal of Financial Markets*, **11**, 228–258.

BORJAS, G. J. (1994), "The Economics of Immigration", Journal of Economic Literature, XXXII, 1667–1717.

- (2003), "The Labor Demand Curve is Downward Sloping: Reexamining the Impact of Immigration on the Labor Market", *The Quarterly Journal of Economics*, **118**, 1335–1374.
- (2006), "Making it in American: Social Mobility in the Immigrant Population" (Working Paper 12088, National Bureau of Economic Research).
- BRONNENBERG, B. J., DUBÉ, J.-P. H. and GENTZKOW, M. (2012), "The Evolution of Brand Preferences: Evidence from Consumer Migration", *The American Economic Review*, **102**, 2472–2508.
- BURCHARDI, K. B. and HASSAN, T. A. (2013), "The Economic Impact of Social Ties: Evidence from German Reunification", *Quarterly Journal of Economics*, **128**, 1219–1271.
- CARD, D. (1990), "The Impact of the Mariel Boatlift on the Miami Labor Market", Industrial and Labor Relations Review, 43, 245–257.

- (2001), "Immigrant Inflows, Native outflows, and the Local Labor Market Impacts of Higher Immigration", Journal of Labor Economics, 19, 22–64.
- CARD, D. and DI NARDO, J. (2000), "Do Immigrant Inflows Lead to Native Outflows?" *American Economic Review*, **90**, 360–367.
- CARR, D. L., MARKUSEN, J. R. and MASKUS, K. E. (2001), "Estimating the Knowledge-Capital Model of the Multinational Enterprise", *American Economic Review*, 91, 693–708.
- CHANEY, T. (2014), "The Network Structure of International Trade", *The American Economic Review*, **104**, 3600–3634. (2016), "Networks in International Trade", in *Oxford Handbook of the Economics of Networks*, Y. Bramoulle,
- A. Galleoti, and B. Rogers (eds.) (Oxford: Oxford University Press).COHEN, L., GURUN, U. and MALLOY, C. (2015), "Resident Networks and Firm Value", *The Journal of Finance*, forthcoming.
- COMBES, P., LAFOURCADE, M. and MAYER, T. (2005), "The Trade-Creating Effects of Business and Social Networks: Evidence from France", *Journal of International Economics*, **66**, 1–29.
- CONLEY, T. (1999), "GMM Estimation with Cross Sectional Dependence", Journal of Econometrics, 92, 1–45.
- CORTES, P. (2008), "The Effect of Low-Skilled Immigration on U.S. Prices: Evidence from CPI Data", Journal of Political Economy, 116, 381–422.
- DA, Z., ENGELBERG, J. and GAO, P. (2011), "In Search of Attention", The Journal of Finance, 66, 1461-1499.

DANIELS, R. (2002), Coming to America (New York, NY: Harper Collins Publishers).

- DUNCAN, B. and TREJO, S. J. (2016), "The Complexity of Immigrant Generations: Implications for Assessing the Socioeconomic Integration of Hispanics and Asians", *ILR Review*, 70, 1146–1175.
- FRIEDBERG, R. (2001), "The Impact of Mass Migration on the Israeli Labor Market", *Quarterly Journal of Economics*, 116, 1373–1408.
- FUCHS-SCHÜNDELN, N. and HASSAN, T. A. (2015), "Natural Experiments in Macroeconomics" (Working Paper 21228, National Bureau of Economic Research).
- FULFORD, S. L., PETKOV, I. and SCHIANTARELLI, F. (2015), "Does It Matter Where You Came From? Ancestry Composition and Economic Performance of U.S. Counties, 1850-2010" (Institute for the Study of Labor (IZA) Discussion Paper No. 9060).
- GARMENDIA, A., LLANO, C. MINONDO, A. *ET AL*. (2012), "Networks and the Disappearance of the Intranational Home Bias", *Economics Letters*, **116**, 178–182.
- GOLDIN, C. (1994), "The Political Economy of Immigration Restriction in the United States, 1890 to 1921", in *The Regulated Economy: A Historical Approach to Political Economy*, C. Goldin and G. D. Libecap (eds.) (Chicago IL: University of Chicago Press), 223–258.
- GOULD, D. M. (1994), "Immigrant Links to the Home Country: Empirical Implications for U.S. Bilateral Trade Flows", *The Review of Economics and Statistics*, **76**, 302–316.
- GREIF, A. (1993), "Contract Enforceability and Economic Institutions in Early Trade: The Maghribi Traders' Coalition", *The American Economic Review*, 83, 525–548.
- GUISO, L., SAPIENZA, P. and ZINGALES, L. (2009), "Cultural biases in economic exchange", *The Quarterly Journal of Economics*, **124**, 1095–1131.
- HEAD, K. and RIES, J. (1998), "Immigration and Trade Creation: Econometric Evidence from Canada", Canadian Journal of Economics, 31, 47–62.
- (2008), "FDI as an Outcome of the Market for Corporate Control: Theory and Evidence", *Journal of International Economics*, **74**, 2–20.
- HECKMAN, J. J. (1979), "Sample Selection Bias as a Specification Error", Econometrica, 47, 153-161.
- HELPMAN, E., MELITZ, M. and RUBINSTEIN, Y. (2008), "Estimating Trade Flows: Trading Partners and Trading Volumes", Quarterly Journal of Economics, 123, 441–487.
- HOLMES, T. J., MCGRATTAN, E. R. and PRESCOTT, E. C. (2015), "Quid Pro Quo: Technology Capital Transfers for Market Access in China", *The Review of Economic Studies*, 82, 1154–1193.
- JAVORCIK, B. S., OZDENC, C. SPATAREANU, M. ETAL. (2011), "Migrant Networks and Foreign Direct Investment", Journal of Development Economics, 94, 231–241.
- JENSEN, E. B., BHASKAR, R. and SCOPILLITI, M. (2015), "Demographic Analysis 2010: Estimates of Coverage of the Foreign-Born Population in the American Community Survey" (Technical report, U.S. Census).
- JUHÁSZ, R. (2018), "Temporary Protection and Technology Adoption: Evidence from the Napoleonic Blockade", *American Economic Review*, forthcoming.
- KATZ, L. F. and MURPHY, K. M. (1992), "Changes in Relative Wages, 1963-1987: Supply and Demand Factors", *Quarterly Journal of Economics*, 107, 35–78.
- KEARNEY, M. S. and LEVINE, P. B. (2015), "Media Influences on Social Outcomes: The Impact of MTV's 16 and Pregnant on Teen Childbearing", *American Economic Review*, **105**, 3597–3632.
- LEBLANG, D. (2010), "Familiarity Breeds Investment: Diaspora Networks and International Investment", American Political Science Review, 104, 584–600.
- LIANG, J. (2017), "Cultural Similarity Measurement using Google Trends" (mimeo University of Chicago).
- MCGRATTAN, E. R. and PRESCOTT, E. C. (2010), "Technology Capital and the US Current Account", American Economic Review, 100, 1493–1522.
- NUNN, N. (2007), "Relationship-Specificity, Incomplete Contracts, and the Pattern of Trade", *Quarterly Journal of Economics*, **122**, 569–600.

- NUNN, N., QIAN, N. and SEQUEIRA, S. (2017), "Migrants and the Making of America: The Short- and Long-Run Effects of Immigration during the Age of Mass Migration" (Working Paper 23298, National Bureau of Economic Research).
- OTTAVIANO, G. I. and PERI, G. (2006), "The economic value of cultural diversity: evidence from US cities", *Journal of Economic Geography*, **6**, 9–44.
- PARSONS, C. and VEZINA, P.-L. (2018), "Migrant Networks and Trade: The Vietnamese Boat People as a Natural Experiment", *Economic Journal*, **128**, 210–234.
- PERI, G. (2012), "The Effect of Immigration on Productivity: Evidence from U.S. States", *The Review of Economics and Statistics*, 94, 348–358.
- PORTES, R. and REY, H. (2005), "The Determinants of Cross-Border Equity Flows", Journal of International Economics, 65, 269–296.
- PUTTERMAN, L. and WEIL, D. N. (2010), "Post-1500 Population Flows and the Long-Run Determinants of Economic Growth and Inequality", *The Quarterly Journal of Economics*, **125**, 1627–1682.
- RAMONDO, N. (2014), "A Quantitative Approach to Multinational Production", *Journal of International Economics*, 93, 108–122.
- RAUCH, J. and TRINDADE, V. (2002), "Ethnic Chinese Networks In International Trade", *The Review of Economics and Statistics*, 84, 116–130.
- RAZIN, A., RUBINSTEIN, Y. and SADKA, E. (2003), "Which Countries Export FDI, and How Much?" (Working Paper 10145, National Bureau of Economic Research).
- REDDING, S. and STURM, D. (2008), "The Cost of Remoteness: Evidence from German Division and Reunification", *The American Economic Review*, 98, 1766–1797.
- SPOLAORE, E. and WACZIARG, R. (2016), "War and Relatedness", The Review of Economics and Statistics, 98, 925–939.
- STEINWENDER, C. (2018), "Real Effects of Information Frictions: When the States and the Kingdom became United", American Economic Review, 108, 657–696.

STEPHENS-DAVIDOWITZ, S. (2014), "The Cost of Racial Animus on a Black Candidate: Evidence using Google Search Data", Journal of Public Economics, 118, 26–40.

STEPHENS-DAVIDOWITZ, S. and VARIAN (2015), "A Hands-on Guide to Google Data" (Working Paper).

- STIGLITZ, J. E. (1990), "Peer Monitoring and Credit Markets", The World Bank Economic Review, 4, 351–366.
- THERNSTROM, S. (1980), Havard Encyclopedia of American Ethnic Groups (Cambridge MA: Harvard University Press).
- VARIAN, H. R. (1990), "Monitoring Agents With Other Agents", Journal of Institutional and Theoretical Economics (Zeitschrift fur die gesamte Staatswissenschaft), 146, 153–174.